MSc Global Development Master's Thesis

Name: Ben Robinson

Student Number: ngz338

Thesis Format: Internship Based Thesis

Supervisor: Sam Jones

Title: Evaluating the Impact of Link Ethiopia's Child Sponsorship Programme: A Mixed Methods Approach

Subtitle: To what extent are child sponsorship programmes an Effective way to Improve Educational Outcomes?

Submission Date: 1st August 2016

Total Keystrokes (Including Spaces): 165,102

Total Standard Pages: 68.8

ECTS: 30







Changing lives through education

Acknowledgements

I would like to thank all of the staff, from both Link Ethiopia and Kindu, in the UK and in Ethiopia for their support in facilitating this research. Without their thoughtful cooperation, from the access of their sponsorship databases to their support with the data collection process in schools, to prompt responses to requests for information, this research would not have been possible. I would also like to thank my supervisor, Sam Jones, for his guidance in both framing the study and support to help overcome challenges posed by the data.

Table of Contents

1. S	ummary	1	
2. Li 2.1. 2.2.	iterature Review Child Sponsorship around the World Measuring Impact in Education in the Global South	5 5 8	
3. Pi 3.1. 3.2.	rogramme Context Education in Ethiopia Child Sponsorship Programmes in Gondar, Ethiopia	.13 .13 .15	
4. Fi 4.1. 4.2. 4.3. 4.4.	ieldwork Methodology and Observations Fieldwork Summary Sampling School Visits Semi-Structured Interviews	.22 .22 .22 .26 .30	
5. Ei 5.1. 5.2. 5.3.	mpirical Strategy Research Questions Outcome Variables Difference-in-Difference Analysis	.32 .32 .32 .32	
6. R 6.1. 6.2.	esults and Interpretation Are Child Sponsorship Programmes an Effective way to Improve Education Outcomes? Do Conditional Cash Transfers, in Addition to the Provision of School Resources, Improve Educational Outcomes?	.43 al 43 53	
7. Di 7.1. 7.2. 7.3. 7.4.	iscussion Relative Effectiveness of the Link Ethiopia Programme Mechanisms of Sponsorship Generalisability of Findings Usefulness of Student-Level Data	.67 67 71 73 74	
8. C 8.1. 8.2. 8.3. 8.4. 8.5.	onsultancy Report Summary of Main findings from the Quantitative Analysis Programme Recommendations Improving Monitoring and Evaluation of the Programme Improving Link Ethiopia Monitoring and Evaluation Conclusions for Link Ethiopia	.76 .76 .80 .89 .92 .93	
Biblio	graphy	I	
Apper	ndix A: Interview Guide – Sponsored Students	. IV	
Appendix B: Interview Guide – Link CoordinatorV			
Apper	ndix C: Interview Ethics	. VI	
Apper	ndix D: Example Regression with Student Fixed Effects	VII	

1. Summary

Child sponsorship programmes represent one of the most visible schemes in the NGO international development landscape to channel small private donations to the Global South. Despite the billions of dollars that pass through these channels, the existing literature analysing the impact of such schemes is surprisingly limited. At the request of Link Ethiopia, a UK-based NGO that operates solely in Ethiopia, this paper analyses the impact of their two parallel child sponsorship programmes, the Link Ethiopia sponsorship scheme and The Kindu Trust (Kindu) sponsorship scheme.

Both schemes support students on the condition that the student is in some form of educational programme. The Link Ethiopia scheme provides school resources and uniforms, access to sponsorship clubs and access to an emergency health and contingency fund to sponsored students, as well as raising money for an 'Educational Environment Improvement Fund' allocated to the school where the student attends. The Kindu Trust scheme provides school resources and access to a 'training fund' to sponsored students. The Kindu scheme also supports the family of the sponsored student, with a cash transfer, reimbursement of medical expenses, medical check-ups, and access to other Kindu programmes.

Both schemes are assessed for their ability to improve the educational outcomes of the sponsored students that are supported through the scheme, although this paper places more emphasis on the Link Ethiopia sponsorship scheme. Given that child sponsorship schemes can vary greatly in nature, in many ways this paper is best placed within the strand of literature that seeks to measure and understand the effectiveness of programmes that seek to improve educational outcomes in the Global South.

Two main research questions are addressed in this paper:

'Are Child Sponsorship Programmes an Effective way to Improve Educational Outcomes?'

'Do Conditional Cash Transfers, in Addition to the Provision of School Resources, Improve Educational Outcomes?'

This paper uses a mixed-methods approach, as it not only measures the impact of the programmes, but also seeks to understand the mechanisms that underlie the sponsorship schemes. It provides recommendations as to how Link Ethiopia can improve the programme, and its monitoring and evaluation processes across the organisation. The research is based on a two-month field research placement at the Link Ethiopia regional office, in the town of Gondar, located in the Amhara region of Ethiopia.

The quantitative analysis uses difference-in-difference analysis on individual-level annual academic data to estimate the programme effects of the two sponsorship programmes, by comparing three groups of students. These are: 98 students sponsored by Link Ethiopia; 71 students sponsored by The Kindu Trust; and 184 non-sponsored students. The time period for valid observations for this study was from the 2005/6 academic year (1998 Ethiopian Calendar), to the 2014/15 academic year (2007 Ethiopian Calendar).

This analysis finds evidence of improved educational outcomes as the duration of sponsorship increases, and this represents a result that is generalisable to many sponsorship schemes that directly support individuals with education. There is little evidence that the programmes improve attainment at school, however there was strong evidence of a relationship between access to, and progress through the education system, and sponsorship duration.

Curiously, there is convincing evidence that there is a non-linear relationship between the duration of Link Ethiopia sponsorship and enrolment and promotion rates, where for the first years of sponsorship these rates decline, but thereafter improve exponentially. There are a number of reasons for this relationship, many of

them idiosyncratic to the Link Ethiopia programme, however they serve to highlight the importance of an effective student selection process in maximising the impact of a sponsorship programme.

This study estimates that at the end of the 2014/15 academic year, the Link Ethiopia sponsorship programme had resulted in an additional 0.31 years of schooling per student, at a cost of £391.53 per additional year of schooling. The efficiency of this programme and positive impact on educational outcomes seem set to improve in the following years. There was no quantitative evidence that conditional cash transfers (CCTs), when present in addition to the provision of school resources improve educational outcomes for students. However, there was qualitative evidence to suggest that CCTs incentivised education.

The qualitative analysis is based on ethnographic observations from the fieldwork, which involved working with the Link Ethiopia and Kindu country staff, visiting schools where sponsored students attend in order to collect the academic data, and through semi-structured interviews with nine sponsored students and one 'Link Coordinator'. The collaboration with staff and schools helped to inform the hypotheses, enhance the reliability, and improve the interpretation, of the quantitative data; and helped to develop potential improvements in monitoring and evaluation for both the sponsorship programmes and across Link Ethiopia's work. The semi-structured interviews found that sponsorship is an effective model to improve educational outcomes, through reducing the private costs to education, providing school uniforms and school resources, and by offering emotional support to raise student's educational aspirations. The interviews also helped to inform

The remainder of the paper is organised as follows. Section 2 provides a literature review of both child sponsorship literature and measuring impact in education in the Global South. Section 3 provides the context of the education system in Ethiopia, and background information on Link Ethiopia, The Kindu Trust and the sponsorship programmes. Section 4 describes the fieldwork methodology and observations. Section 5 outlines the empirical strategy employed on the quantitative data. Section

6 presents the results from the quantitative data and interprets these findings. Section 7 places the quantitative findings back into the broader context of the effectiveness of educational programmes in the Global South, and uses the semistructured interviews to explore the mechanisms that allow sponsorship programmes to be effective. Section 8 is a consultancy report, primarily targeted at Link Ethiopia, which summarises the main findings, and presents possible recommendations to improve both the sponsorship programmes, the monitoring and evaluation of the programme and potential future research to be carried out on the programme, before offering ways in which Link Ethiopia can improve its monitoring and evaluation across its Ethiopian programmes.

2. Literature Review

2.1. Child Sponsorship around the World¹

Child sponsorship is a model of north-south private financial flows, one that varies from NGO to NGO; however programmes typically include regular giving, a desire for donations to primarily benefit individuals and the provision of regular updates to sponsors (Watson & Clarke 2014). Child sponsorship may be to some organisations little more than an effective fundraising mechanism where donations are 'given in the name of a sponsored child...to create village-level public goods' (Wydick et al. 2013: 397), whereas in others, child sponsorship is 'programmatically defining as an individualised model of development' (Robinson 2016: 9).

Organisations such as Plan International and Save the Children favour the former model, whereas at Compassion International 'sponsors pay for children's school tuition and uniforms, several nutritious meals per week, health care, and tutoring' (Wydick et al. 2013: 395). This individualised model of development is perhaps the most recognisable model of child sponsorship, particularly in the eyes of the donor, and for Rabbitts (2012) child sponsorship 'usually takes the form of long-term commitments to monthly remittances that cover the medical and educational requirements of individual children in the Global South'.

Proto-models of child sponsorship were arguably pioneered by Save the Children UK and the Society of Friends in Austria as early as 1920, providing rations to children (Watson 2014b). Organisations built on the initial successes of these programmes. Schemes that 'involved direct handouts, service provision and cash transfers to children and their families for food, medicine and school needs' (Watson 2014a) grew particularly in the 1960s and 1970s. In the 1980s criticism of child sponsorship programmes in an international development context grew, and these arguments were crystallised in the famous 'Please do not sponsor this child' (Stalker 1982) article. The article criticised sponsorship programmes for high operational costs, favouring certain individuals over others in deprived communities rather than

¹ This section draws heavily on the background material found in Robinson (2016) – See Section 4.1 Child Sponsorship Models

supporting community development, and for cultivating empty aspirations that would never be realised for beneficiaries.

This shift in attitudes prompted the emergence of the village-level public goods model of child sponsorship as a fundraising mechanism for community development programmes, as these were considered to be schemes with greater potential for sustainable poverty reduction than individualised giving (Watson et al. 2014). However, in recent times, there has been a re-evaluation of whether such criticism is completely warranted. Whilst sponsorship is still not considered a solution for sustainable or systemic poverty reduction, 'if designed wisely, targeted carefully, and implemented efficiently, sponsorship programmes usefully add to the livelihoods of disadvantaged families in environments that lack alternative safety nets' (van Eekelen 2013: 474).

Whilst the variety of child sponsorship programmes makes it difficult to define what is, and what is not a sponsorship scheme, when including both the public goods and individual support models, an estimated \$3.29bn (Wydick et al. 2013) is thought to flow through child sponsorship schemes annually. Given the large volume of monetary flows through child sponsorship programmes, the amount of literature measuring the impact of these programmes is surprisingly limited. Until 2013, the only example of a widely available impact evaluation was a randomised control trial (RCT) analysed by Kremer et al. (2003), which measured the impact of the provision of classroom construction, school uniforms and textbooks through a sponsorship model conducted by a Dutch NGO, International Christelijk Steunfonds (ICS). Such a programme could be considered to be both a combination of the public-goods model (classrooms) and individual support (uniform and textbooks). This paper found that these interventions had a positive impact on attendance at school, and reduced dropout rates, resulting in a 17% increase in years of schooling, and improved grade attainment (grade progressed in the time period), by 15%.

As programmes, models of individual support, and public goods models are very different, and therefore measuring the impact of the two different models would require two very different approaches. Wydick et al. (2013: 397) suggest that the

potential impacts of programmes that use the village-level public goods models are 'more diffuse and more difficult to assess'. In light of this issue, when carrying out an impact assessment of Plan International's sponsorship scheme, Pettit & Shutt (2008:13) opt for a 'comparative analysis of Plan's own intentions, practices and outcomes, rather than [] a test against a specific set of standards or indicators'.

In the aforementioned paper, Kremer et al. (2003) are able to overcome this challenge for a number of reasons. Firstly, the analysis was carried out as a prospective RCT, with treatment and control schools, meaning that the study did not have to be concerned with other factors influencing school performance due to the treatment schools being randomly selected. In the case of the Plan International analysis, any assessment would have been both retrospective, and the distribution of public goods non-random, meaning any analysis would be subject to omitted variable bias. Secondly, the ICS sponsorship scheme was designed such that the vast majority of students received free uniforms, and the textbooks were distributed to the schools to be used by students. This meant that the effect of the programme, could be measured at a school-level, with the entire package of uniforms, textbooks and classroom construction treated as a school-level public good, rather than measuring the impact of individual support for a child, at an individual level.

This lack of research estimating the impact of individualised models of child sponsored changed with Wydick et al.'s (2013) paper, *'Does International Child Sponsorship Work? A Six-Country Study of Impacts on Adult Life Outcomes*'. This represented a comprehensive evaluation of a large-scale child sponsorship programme, analysing data 'on the life outcomes of 10,144 individuals...from six developing countries that are representative of the Compassion program's work worldwide: Bolivia, Guatemala, India, Kenya, the Philippines, and Uganda' (Wydick et al. 2013: 394).

Wydick et al.'s (2013) paper examines the impact of the programme on the educational, employment and community leadership outcomes of sponsored students. In terms of educational outcomes, the outcome variables of interest to Wydick are total years of formal education, and the percentage of students

completing primary, secondary, and university education. The study uses both OLS and IV regressions, with fixed-effects at a household level, and estimates for spillover effects, both within, and between households, to estimate the impact of the Compassion sponsorship programme on these outcome variables. The paper ultimately finds that the Compassion sponsorship programme has positive impacts on all of these outcomes, and suggests that the focus of the Compassion programme on building the self-esteem and aspirations of the students is the driving factor of these positive outcomes. Glewwe et al. (2013) confirm this causal relationship between the Compassion sponsorship programme and improved selfesteem, and higher expectations for educational and employment outcomes.

This, and the Kremer et al. (2003) paper remain the only two, widely available quantitative studies that measure the impact of child sponsorship. However, as Wydick et al. (2013: 395) indicates, 'Compassion [sponsorship] projects are similar to many government and international donor programmes that promote education. Sponsors pay for children's school tuition and uniforms, several nutritious meals per week, health care, and tutoring.' Keeping this in mind, studies that explore effective ways to improve educational outcomes in the Global South represent a broader strand of literature that child sponsorship impact evaluations can be considered to be part of.

2.2. Measuring Impact in Education in the Global South²

Despite criticism from esteemed economists such as Deaton (2010), RCTs are often seen as the 'gold standard' when measuring the effectiveness of development programmes. Glewwe et al. (2004) illustrate the advantages of prospective over retrospective analysis by carrying out different analytical approaches to measure the impact of introducing flipcharts to schools in Kenya. They demonstrate how that even when controlling for observable school inputs, results from OLS regressions can suggest that the introduction of flipcharts increases test scores by 20% of a standard

 $^{^{2}}$ This section draws on the background material found in Robinson (2016) – See Section 5.1 How do we measure impact in education?

deviation. However, when the same experiment is carried out as an RCT, there is no evidence that flipcharts have a statistically significant impact on test scores.

They attribute this difference to a failure of the retrospective methodology to adequately account for omitted variable bias, and thus upwardly biasing the programme effects. This occurs if 'it is more plausible that variation in the input of interest is influenced by variation in the unobserved inputs, or that unobserved inputs vary in response to the (observed) input of interest' (Glewwe et al. 2004: 256). Whilst this means that prospective analysis is generally preferable, clearly the possibility to do so does not always exist. The study also shows that when retrospective analysis is used, the results from a difference-in-difference analysis, whilst still overstating the impact of flipcharts compared to the prospective analysis, are much closer to the prospective analysis than a mere OLS regression. Thus, if a difference-in-difference approach is used, and awareness of possible upward bias is maintained, this may represent a best-case scenario when retrospective analysis is necessary.

Irrespective of the method, understanding the results from different impact assessments of educational programmes provides important context, allowing different programmes to be compared in their effectiveness. Common to such studies, is an understanding of education as a system that takes in a range of inputs, to produce educational outcomes, and then an attempt to measure the effect of a change in these inputs. Across the literature, this is done by a construction of counterfactual outcomes (outcomes in the absence of the programme), which allows the isolation and estimation of the change in educational outcomes due to a change in a given input or inputs.

Given the preference in the literature for RCTs, many studies use this methodology. Often quoted examples of these studies include Evans et al. (2009), which shows that the distribution of free school uniforms reduced absenteeism by 39% across all students, and by 66% when only considering the poorest students; and Kremer et al. (2009) who find that a merit-based scholarship in the form of school fees and grants for girls in Kenya improved attendance by 5% and possibly caused small improvements in test scores for both boys and girls. Alternatively, Glewwe et al.

(2009) find that textbook provision has little effect on test scores, except for the best performing students. As well as educational inputs, health and nutrition inputs may also have an effect on educational outcomes, as Vermeersch & Kremer (2004) found that the provision of school meals can increase school participation by 30%.

One form of programme that has received more extensive analysis is the impact of conditional cash transfers (CCTs), where families receive cash in exchange for the attendance of their children at school, on educational outcomes. The best known, and most-evaluated of the schemes (due to it being designed as an RCT) is the PROGRESA programme in Mexico, now known as Oportunidades. The programme resulted in higher enrolment rates, lower grade repetition, and lower dropout rates (Behrman et al. 2005), with further analysis estimating that receiving the cash transfers for 5.5 years increased grades completed by 0.8–1.0 years (Behrman et al. 2011).

Despite the preference for RCTs there are also a range of studies that use retrospective analysis to assess the impact of different inputs on educational outcomes. One example is Filmer & Schady's (2008) analysis of an annual \$45 scholarship for girls in Cambodia, which increased both enrolment and attendance by about 30 percentage points, with greater benefits for girls with lower socio-economic indicators. In another study, Handa & Peterman (2007), find that child attainment in South Africa is partly a function of their nutritional status, Kingdon & Drèze (2001) demonstrate that the provision of midday meals, as means to reducing the private costs of schooling, increases the chance of a girl completing primary education by 30%, and improves their attendance by 15%.

In recent years, the measurement of impact in development programmes has been influenced by effective altruism. Effective altruism is best understood as a movement constituted of academics, think-tanks, such as the Centre for Effective Altruism, and non-profit organisations, such as GiveWell, that encourage individuals to donate more of their income, but only to those causes that have the most effective outcomes. MacAskill (2015) and Singer (2015) combine the moral philosophy of utilitarianism with the growing evidence base of econometric analysis to make the

case for increases in foreign aid, particularly through private channels. At the same time, they argue that given that the financial resources for foreign aid are finite, the distribution of aid should be directed towards those programmes that have the biggest impact per dollar expended, meaning that cost-effectiveness is a more relevant consideration than merely an aggregate measure of programme effects.

In the field of education in the Global South, however, such evaluations are limited in number, and 'little reliable information is available regarding how to improve school attendance and school quality, or on the true relationship between schooling and later-life outcomes such as income' (GiveWell 2009). Some exceptions to this are Miguel & Kremer (2004), who calculate that a deworming programme can deliver an additional year of schooling for \$3.50; Evans et al. (2009), who calculate that a uniform distribution programme costs \$91 for each additional year of schooling, and Evans & Ghosh (2008), who calculate that the PROGRESA programme costs \$614 for each additional year. Whilst the first two studies both take place in a Kenyan context, the PROGRESA programme of course operates in a Mexico, a middle-income country, which at least in part explains the higher costs to improving educational outcomes.

Whilst a retrospective study is carried out on a limited set of already available data, as opposed to a prospective study where the data collection is planned prior to the intervention, the metrics used in assessing the impact of educational programmes are largely similar across the literature. Educational impact assessments often refer to 'increased years of schooling'. In the deworming study Miguel & Kremer (2004) estimate years of schooling gained through improved attendance, whereas Schultz (2004) calculates the increased years of schooling on the basis of increased enrolment rates under the Mexican PROGRESA programme. The use of such standard metrics facilitates the easy comparison of the effectiveness of different programmes.

'Increased years of schooling', however, is not the only relevant metric when measuring educational programmes. In the examples mentioned so far, attendance rates, enrolment rates and test scores have all been relevant outcomes to measure.

As already mentioned, in Wydick et al.'s (2013) analysis of the Compassion child sponsorship scheme, the chosen metrics are the completed years of schooling, and completion rates of primary, secondary and university education.

Many impact evaluations not only account for the impact of the programme on the targeted beneficiaries, in this case the sponsored children, but also account for any spillover effects. In identifying the impacts of a deworming programme in Kenya, Miguel & Kremer (2004) measure the impact of the treatment at a school level to account for externalities between students, and the impact on neighbouring schools for inter-school spillovers. Likewise, Wydick et al. (2013) estimate the effects of sponsorship on younger siblings.

Circumstances though, require that in this paper historic data is used, and that retrospective analysis is required. This paper therefore uses a difference-indifference approach similar to that used in Glewwe et al. (2004), an approach that is elaborated on in the 'Empirical Strategy'.

3. Programme Context

3.1. Education in Ethiopia³

The backdrop to education in Ethiopia is the prevalence of poverty, with 30% of Ethiopians living in extreme poverty (based on a national poverty line of \$0.6 a day). This has however, fallen from 39% since 2005 due to strong recent economic growth (Ministry of Education 2015a). Despite spending 25.2% of government expenditure on education in 2013/14 (Ministry of Education 2015b), this still only amounted to 39,857 million Ethiopia Birr (ETB) (\approx \$1,811 million) (Ministry of Education 2015b), approximately only \$20 per capita, a very low level by international standards. Figure 1 illustrates this lack of finance for education in Ethiopia, and its dependence on foreign aid, as it is facing a financing gap of 70,131 million ETB over the next five years.

	2015/16	2016/17	2017/18	2018/19	2019/20	Total
(a) Total cost	73,635	81,350	91,831	98,781	108,225	453,822
(b) Gov't budget	58,838	66 <mark>,</mark> 944	76,102	86,445	95,362	383,691
Gap (a-b)	14,796	14,406	15,729	12,336	12,863	70,131

Figure 1: Estimated financing gap for each year of ESDP V (million birr) Source: (Ministry of Education 2015a)

In spite of these challenges though, in working towards the Millennium Development Goals (MDGs) Ethiopia has seen significant progress in increasing its Net Enrolment Rate at primary school from 82% in 2009/10 to 93% in 2013/14 (Ministry of Education 2015b), suggesting that enrolment in the early stages of schooling is not the greatest challenge the Ethiopian education system faces. Given improvement in this area, the subsequent challenge facing Ethiopia is to address the high dropout and repetition rates, and to improve the progression of students through the education system (UNESCO 2015). Dropout and grade repetition is high in the early

³ This section draws heavily on the background material found in Robinson (2016) – See Section 4.2 Education in Ethiopia

grades, and therefore completion of primary school remains low, with only 69.5% of students completing Grade 5, and only 46.7% do likewise for Grade 8. These rates of completion have fluctuated up and down on an annual basis, and in fact indicate an decreasing trend over the past five years (Ministry of Education 2015b).

The issue of dropout and repetition appears to be endemic to the system, and government targets of 1% rates of dropout and repetition from ESDP IV have been missed by a large margin. Repetition rates remain at about 8% (Promotion = 92%), and dropout rates in Grade 1 are 22%, and 11% thereafter (Ministry of Education 2015a). The Ethiopian governments attributes these high rates of dropout to 'low quality educational inputs, i.e. skilled teachers, relevant teaching and learning materials etc.' (Ministry of Education 2015a: 19). A major quantitative study of the causes of dropout in Ethiopia identified a range of school-level factors that impact on the dropout rate including pupil-teacher ratios, textbook-pupil ratios, and the ratio of teachers with teaching qualifications. The study recommends improving these ratios at a school-level, but also 'developing further guidance for schools to identify and support children at risk of dropping out while they are still in school' (Jennings & Poppe 2012: 60), suggesting that the circumstances of individual students are as big a determinant of dropout as school-level factors.

Access to education, is not evenly distributed, in particular with respect to geography (rural/urban), gender and socio-economic differences. ESDP V (Ministry of Education 2015a: 36) states that after Grade 4 'those of lower socio-economic status do not have equal access to education', citing financial and geographic barriers. There has however, been significant progress in closing the gender gap, particularly in relation to primary education where both enrolment, dropout and completion rates are all very similar for both genders (Ministry of Education 2015b). This transformation though has not translated to secondary education, where a gender gap still exists and widens (Ministry of Education 2015b; Gretland et al. 2014).

3.2. Child Sponsorship Programmes in Gondar, Ethiopia

3.2.1. About Link Ethiopia⁴

Link Ethiopia is a small education charity, which began existence as a series of informal projects in 1996, but has been a registered charity since 2006. The head office is based in London, UK. The charity operates in Ethiopia, with regional offices in Gondar and Bishoftu (Debre Zeit), and its programmes run across the Amhara and Oromiya regions of Ethiopia. In 2014/15 Link Ethiopia had an income of £488,769 and expenditure of £476,357 (Link Ethiopia 2015a).

In 2016, Link Ethiopia experienced a major change in the organisation, in that it formally 'merged' with another UK-Ethiopian education charity, 'The Kindu Trust', under one governance structure in the UK. Previously the two charities had worked very closely together, and as things stand in Ethiopia, the organisations still run in parallel, despite now sharing several resources such as staff and office space, although an Ethiopian merger will also be complete in the near future.

According to Link Ethiopia's own mission statement, '*Link Ethiopia changes lives through education. We work in partnership with local communities to improve access to quality education for all students in Ethiopia, encouraging mutual understanding and respect between different cultures*' (Link Ethiopia 2016). The organisation has a range of programmes to this end, including improving school infrastructure, facilities and resources in Ethiopian schools, teacher training, and schemes that promote inclusive education by improving access to education for marginalised groups, namely girls, young people with disabilities, and rural students (Robinson 2016). In every large school, one teacher is appointed as a 'Link Coordinator', who, along with the school director, is the main point of contact in the school for Link Ethiopia, and helps to facilitate Link Ethiopia projects in the school.

One of Link Ethiopia's major programmes to increase access to education is a child sponsorship programme. The programme seeks to identify children who without

⁴ This section draws heavily on the background material found in Robinson (2016) – See Section 2.1 About Link Ethiopia

additional (primarily in-kind financial) support would not be able to continue their education, and seeks to prevent their dropout from school.

3.2.2. Link Ethiopia Sponsorship Model⁵

As a response to both the criticisms of individual models of child sponsorship, and the recent re-evaluation of those criticisms, Link Ethiopia's model combines aspects of both the individual and public goods models of sponsorship, in what can be termed a 'hybrid model'. Link Ethiopia has three categories of sponsorship, 'standard', 'bespoke' (where donors support a specific individual with additional cash transfers), and 'further' (where an individual is sponsored through tertiary education). Given that out of a total of 197 sponsored students, only 17 are categorised as bespoke sponsorships and 3 are further sponsorships, this impact assessment only concerns the standard sponsorship model. Whilst Link Ethiopia sponsors students in both the Amhara and Oromiya regions, this study only concerns the Amharan programme, based out of the Gondar office, which represents just over half the total students sponsored through Link Ethiopia.

Link Ethiopia describes its sponsorship programme as follows:

"sponsorship is the provision of support to an intended beneficiaries' educational attainment, achievement, and progress and school attendance. It is a conditional assistance to enable children or adults who are at risk of dropping out from education, to enable them continue their education to the best of their ability, and within the limitations of the LE sponsorship programme." (Link Ethiopia 2015b)

Figure 2 outlines the budgeted financial expenditure of the programme, where each sponsor pays £15 per month. 16.67% is spent on sponsor feedback and operational costs. Of the remaining budget 45% is spent on the educational environment improvement fund. This represents the public goods aspect of the model, as this money is used to improve the school that the child attends through other Link Ethiopia programmes such as classroom building, WASH, and literacy projects.

⁵ This section draws heavily on the background material found in Robinson (2016) – See Section 4.3 Link Ethiopia's model

12



Changing lives through education

Sponsorship Guidelines

Financial Model

£15 (pounds) per month for a beneficiary paid by the sponsor.

The standard sponsorship budget per student is:

Ar	ea of spending	Annual budget	Percentage
	Sponsor feedback and operational costs	£30	16.67%
	Charitable spending (sub divided below)	£150	83.33%
Total budget		£180	100%

The charitable spending budget is sub divided as follows:

Ch	aritable spending	Annual budget	Percentage	
	School clothing and resources fund	£45	30%	
	Educational environment improvement fund	£67.50	45%	
	Sponsorship club fund	£30	20%	
	Health and contingency fund	£7.50	5%	
То	tal charitable spending budget	£150	100%	



Figure 2: Source: (Link Ethiopia 2015b)

The corresponding 55% is therefore spent on educational support for the individual students that are supported by the sponsorship programme. This is split between school resources and clothes, sponsorship clubs, and an emergency health and contingency fund. The focus on education is clear, with only 5% of charitable spending allocated for emergency health and contingency, with no money available for routine health expenditure. School clothing and resources includes school uniform, a school bag, shoes, stationery, and exercise books to help reduce the private financial cost of education. Sponsorship clubs are also only attended by direct beneficiaries of the programme, designed to provide a safe space for students to spend time with one another, gain confidence and improve social skills.

This model was adopted and first implemented in the 2012/13 academic year. Prior to that Link Ethiopia had more heavily favoured the public goods model, with 80% of charitable spending being allocated towards the educational improvement fund, and the remaining 20% being spent on school clothing and resources. Sponsorship clubs have only been implemented in the Amhara region since the 2014/15 academic year. Up until 2013, the monthly contribution by a sponsor was £12 per month, but this figure was changed to £15 to compensate for inflationary pressures in Ethiopia. In 2014/15 the Amharan sponsorship scheme had a total expenditure of £13,821.

As a purely educational programme, opposed to a poverty reduction scheme or a holistic child sponsorship model, the programme can also be viewed similarly to 'scholarship' schemes, however the term scholarship implies that selection for the scheme includes educational ability as one of the criteria. In contrast, Link Ethiopia selects students for its standard scheme on a 'needs' basis. An effective selection process is likely to be a key determinant of the success of the programme, as the premise of the programme is that without additional support the selected students would dropout from school. Link Ethiopia identifies students as shown in Figure 3. Whilst at the advent of the scheme, students were recommended for sponsorship by the schools that the scheme operates in, as of 2014, due to the introduction of new government regulations, only students on a list (of deprived families) were eligible for sponsorship.

Identification & Recommendation for Sponsorship

Based on school assessment, Ethiopian Women & Children's Office recommendation and Sponsorship Manager assessment, children can be identified as in need of support.

The identification and recommendation of a child or young adult will be performed by the Sponsorship Manager (with support from other staff members when necessary), and the school (with reference to Ethiopian Women & Children's Office advice/lists). These beneficiaries need to be identified by the school/learning institution in conjunction with Link Ethiopia according to the following criteria:

- Children and young adults, 4 to 18 years of age.
- Children and young adults who are currently enrolled in and attending fulltime education, and students who are or were attending full-time education but are currently struggling to cope with school and demands at home, e.g. working.
- Score on "Baseline Survey Simple Poverty Scorecard" document.

Priority shall be given to: Single or double orphaned children who are identified as vulnerable and in need of support by the school/learning institution in conjunction with Link Ethiopia.

Gender: Considering the difficult circumstances facing girls' attempts to access education, Link Ethiopia has set a minimum limit on the number of female sponsorships. This number is set at 75% or two-thirds.

Figure 3

Source: (Link Ethiopia 2015b)

3.2.3. About The Kindu Trust

The Kindu Trust was founded in 1998 and also has its head office in London. In 2014/15 it had an income of £124,533 and an expenditure of £147,451 (The Kindu Trust 2015), although unlike Link Ethiopia, its work is solely focused in the Amhara region. For its first four years of existence The Kindu Trust existed as a series of residential homes, taking 80 vulnerable children into care, however, after four years the residential homes were closed so that the children could live with relatives or be

placed with foster families. Instead of funding residential homes, the scheme was transformed into the current Child Sponsorship Programme (The Kindu Trust 2016b).

Also, unlike Link Ethiopia, the child sponsorship programme is the centrepiece of The Kindu Trust's work, however, it does also have a range of community-based projects, such as a playgroup and income-generating projects, which sponsors and their families may also benefit from.

3.2.4. The Kindu Trust Sponsorship Model

The Kindu Trust Sponsorship model is primarily a model of conditional cash transfers for attendance in school. The family of the sponsored child receive 300 ETB a month (approx. £10), conditional on the student being enrolled in school. This cash transfer increases to 330 ETB for students over the age of 18. The Kindu Trust currently supports over 350 children in this way. The scheme can be seen to be similar programmes such as PROGRESA, where cash grants to families were conditional on children attending 85% of school days (Schultz 2004), however the sponsorship is not conditional on set attendance rate, but rather a judgement call of whether that student is attending education or not.

The programme though is more than just a CCT, in that there are other benefits to the family other than just cash, benefits which are still conditional on the sponsored student's being in school. These benefits include school resources, including uniforms and exercise books (but not shoes and a school bag as with Link Ethiopia), medical check-ups for the entire family, reimbursement for medical expenses, and contributions to a 'training fund', which beneficiaries can apply for to take vocational courses (The Kindu Trust 2016a). The total cost of the sponsorship programme to the donor is £19 per month.

Compared to the Link Ethiopia scheme, the Kindu sponsorship programme is a more holistic sponsorship model, with greater expenditure on health and direct transfers, but with less expenditure on school resources than Link Ethiopia. There are also greater benefits to the family of the sponsored child, than just reduced costs to education, as they also benefit from the cash transfer and medical support. The cash transfer is seen to be an incentive for a household to send their child to school, but to

also have real benefits to the family. When considering the impact of the programme, it is clear that the impact on educational outcomes are not the only benefits, but for this evaluation, only the impact on educational outcomes is being considered.

Students for the sponsorship programme were initially selected based on the judgement of the Ethiopian social workers, however, like Link Ethiopia, since 2014 students have been selected from lists from the Women's and Children's Office and a home visit assessment. The Kindu Trust gives priority to children who are disabled, orphaned, are HIV positive, or have parents who are HIV positive (The Kindu Trust 2014).

Resource Provided	Link Ethiopia	The Kindu Trust
Learning Resources	Yes	Yes
Uniform	Yes	Yes
School Bag and Shoes	Yes	No
Cash Transfer	No	Yes
School Fund Contributions	Yes	No
Sponsorship Club	Yes	Yes**
Medical Expenses	Yes*	Yes
Family Health Checks	No	Yes
Training Fund	No	Yes

 Table 1: Comparison of Link and Kindu Sponsorship Programmes

* Only for medical emergencies

** Access to a separately funded after-school club

4. Fieldwork Methodology and Observations

4.1. Fieldwork Summary

The data for this study was gathered through a two-month field study based in the town of Gondar, in the Amhara region of Ethiopia. Based out of the Link Ethiopia Gondar office I worked with different members of the staff team, including the Sponsorship Coordinator to visit schools where sponsored students attend in order to collect academic data on students. These school visits formed the basis of the fieldwork.

A mixed-methods approach has been taken, combining quantitative analysis with ethnographic observations from my time working with Link Ethiopia and visiting schools and semi-structured interviews with sponsored students and school link coordinators. In measuring the effectiveness of programme, greater weight is given to the quantitative analysis, although qualitative data has helped to inform the hypotheses to be tested and the interpretation of the data rather than using the qualitative data as a free-standing piece of analysis into the impact of the programme.

The quantitative analysis is found in the 'Empirical Strategy', and 'Results and Interpretation' sections. The qualitative data features in this section under 'School Visits', and the semi-structured interviews are analysed in the 'Discussion', providing evidence for the mechanisms of sponsorship, and in the 'Consultancy Report' informing programme recommendations.

Whilst the Kremer et al. (2003) analysis of the ICS programme was able to measure the impact of both individual support and public goods, the number of students supported by Link Ethiopia and Kindu does not constitute a high enough proportion of students in a school to measure the impact at a school-level. Therefore, this study only measures the impact of the individual educational support to students.

4.2. Sampling

This impact assessment takes the individual as the unit of investigation, in that it seeks to measure the impact of the individual educational support of the programme

on the individuals participating in the programme, as opposed to any wider benefit to the school, families or local community. With this in mind the sample for the study is drawn from schools and colleges where Link Ethiopia currently has sponsored students attending, as the sampling decision had to be based on information available prior to starting the fieldwork, and often only the current school of the student was known.

The study can be seen to consist of three groups, Link Ethiopia sponsored students, Kindu sponsored students, and non-sponsored students, all of which attend or attended the schools where Link Ethiopia students currently attend. With such a sampling frame all schools that were part of the study were government state schools or colleges.

Students who receive bespoke sponsorship i.e. support above the standard support of either the Link or Kindu scheme were removed from the study. A small number of students who receive additional funds because of health problems such as HIV/AIDS are still included in the study, as such support was merely designed to mitigate the costs and challenges of this for the student, rather than as extra support towards their education. Likewise, if a student did require extra support, for example to pay school fees for a government college, but had been on the standard sponsorship model prior to this, they were still included in the study. Further details of the sampling for both the quantitative data and semi-structured interviews can be found below.

4.2.1. Quantitative Sampling

As mentioned, the initial sampling frame for the study was the schools in which Link Ethiopia students, who were supported through its standard sponsorship model were currently attending, giving a total of 18 schools.

Once the 18 schools were selected the internal Link Ethiopia database of sponsored students was used to identify the students that were attending, or had previously attended these schools. This produced an initial sample of 119 students, of which 6 were removed as they had received bespoke sponsorship and a further 15 students were removed when no academic data was found, leaving a final sample of 98

students sponsored by Link Ethiopia. The sample of 98 students included not only current students, but also students who were no longer on the scheme, either because they had dropped out, moved schools, or no longer required sponsorship. Information was drawn from the database for each student, including date of birth, gender, start date of Sponsorship and end date of sponsorship (if applicable).

For the Kindu students, the only list that included the information of the school currently attended was a list of current beneficiaries attending school. There was an initial sample of 121 students attending the 18 sample schools, with 6 students then removed from the sample for being bespoke sponsorships, and then a further 44 students removed when no data was found, leaving a final sample of 71 students sponsored by Kindu. Data was matched from a corresponding database to gather the same background data on Kindu students as available for Link students.

The final group of students in the quantitative study are the non-sponsored students. The non-sponsored students are randomly selected from the 18 schools visited. For half of the primary schools, five students were randomly selected from Grade 1 in each of 2000, 2002 and 2004 (Ethiopian Calendar [E.C.]), (2007/08, 2009/10 and 2011/12 Gregorian Calendar [G.C.]), and in the other half of the primary schools 5 students were randomly selected from Grade 1 in each of 2001, 2003 and 2005 (E.C.) (2008/09, 2010/11 and 2012/13 G.C.). In secondary schools 5 students were randomly selected from Grade 9 in 2004, 2005 and 2006 (E.C) (2011/12, 2012/13 and 2013/14 G.C.). This sampling method was used to ensure that observations were captured across a range of years in the 10-year study period, and to include students that are currently of different ages and grades, whilst ensuring that there were large number of observations (a Grade 1 student in 2007 E.C. (would only have had one possible observation), to ensure that the random sample was robust. Age and gender data from these students was taken from the school rosters.

Some schools were less willing to give their support to tracking randomly selected students, and in schools where record-keeping was poor and there were a large number of students tracking randomly selected students was very difficult. In certain schools therefore, random students' data was not collected, and in schools where

random students' data was collected, this was sometimes for less than 15 students. In total, data was collected on 184 randomly selected students. There was at least one observation for all non-sponsored students, so all 184 students remained in the sample.

4.2.2. Qualitative Sampling

Semi-structured interviews took place with sponsored students of both Kindu and Link Ethiopia. Initially it was also planned to carry out interviews with students who had dropped out and left the scheme, however, communication was generally lost with these students so they were unable to be contacted for interviews. Nonsponsored students at the schools were to be interviewed to understand their view of sponsorship schemes, however interviews with sponsored students suggested that very few non-sponsored students would have been aware of the Link Ethiopia an Kindu programmes, and therefore these interviews were considered of a lower priority, and not realised due to time constraints.

Five Kindu and four Link Ethiopia students were interviewed, and the students were selected to cover a range of ages, schools, grades, and sponsorship experiences. Two Link Ethiopia primary school students were interviewed, a preparatory school student and a student at a government polytechnic college. Of the five Kindu students, one primary school student was interviewed, one secondary school student, one preparatory school student, one polytechnic student and one university student. Not all of the students selected ended up in the final data set due to not finding their historic academic data on school visits, but with the exception of the Kindu University student, all students were part of the initial data set drawn from the databases. Finally, a Link Coordinator from a school who was familiar with the sponsorship programme and sees the children regularly was interviewed. A summary of the students interviewed is shown in the table below.

Initials	Age	Gender	Grade	School	Years of Sponsorship	Organisation
TE	19	Μ	11	Fasilides Preparatory	7	Link
FT	16	F	TVET	Gondar Polytechnic	7	Link
EA	17	F	7	Felege Abyiot	2	Link
MM	11	F	5	Atse Bekafa	3	Link
FG	24	М	TVET	Gondar Polytechnic	15	Kindu
МВ	11	F	6	Arbartu Ensesa	6	Kindu
SA	16	М	11	Fasilides Preparatory	11	Kindu
AE	16	М	9	Angereb	9	Kindu
НВ	23	Μ	University	University	15	Kindu

Table 2: List of sponsored students interviewed

4.3. School Visits

4.3.1. Observations

The visits to the 18 schools in the study formed the bulk of the fieldwork, as these visits were used to collect the historic academic data on the students in the sample. They involved working with a Link Ethiopia staff member as a translator, and then working through the academic records with the school record keeper to find the historic data on the sample students. By working with the record keepers and Link Ethiopia staff I was able to gain a better understanding of the outcomes of students, especially as the records were not always definitive and so required collective interpretation was required.

In most schools, there was a designated record keeper employed at the school, with historic school rosters, sometimes dating back over 30 years, and transcripts and

certificates of current students kept in a separate designated 'record room'. Only in small schools, such as Andinet Elementary, which only served grades 1-4, was this lacking, in which case the school director was responsible for the record-keeping and records were kept in the director's office. Access to the records was negotiated between Link Ethiopia staff and the school director, as Link Ethiopia already has good working relationships with the schools involved in the study, and it was agreed that photos of rosters and transcripts of students could be taken for use in the study, to then be entered into the dataset.

The role of the record keepers in the school is to manage the record room, keeping records organised, and to prepare official documentation such a certificates and transcripts. They are also on hand when teachers, students or parents have queries about previous results. The rosters, which provided the bulk of the data, are the responsibility of the class teachers to fill out, whilst the record keeper merely organises them, and this perhaps explains a lack of consistency not only between schools but between classes. This also means that except at peak times, such as exam time, there is spare capacity in the role of the record keepers. At the same time, record keepers suggested that data collection of this type was extremely unusual and represented a significant increase in their workload. Reactions to this increased work were varied, from engagement with the research and questions as to the purpose of the data, to a reticence to help and demands for additional payments, which were refused.

4.3.2. Collection of Quantitative Data

The rosters, an example of which can be seen in Figure 4, were largely standardised between schools, suggesting they were from a government or local authority template, which could vary slightly from school to school or class to class. All rosters contained space for the student's name, age, gender, their promotion to the next grade or their being 'detained' and having to repeat the year, and then the internal exam results for each subject taken, overall average, and days absent for the first semester, second semester and average/total for the year. In secondary schools these records were kept in English, and tended to not collect data on days absent,

and in primary schools, records were kept in the regional/national language, Amharic.

Age data was nearly always kept, and the age listed was assumed to be the age of the student on the first day of the Ethiopian calendar year (12th September) as this represents the start of the school year. Given that dates of birth are often unknown in Ethiopia it made little sense to attempt to be more precise in the age of students than this. Results from internal exams were recorded for each subject, however the subjects taken varied from year to year, and occasionally by school, and this level of detail was not required, so an average of all exam results for the year was recorded as measure of attainment.



Figure 4: A secondary school roster

Whilst guidelines for promoting or detaining a student based on internal exam results do exist, ultimately the decision was at the discretion of the school, as evidence of the guidelines being ignored was found. At the end of primary school (Grade 8), students take regional exams, which they must pass in order to proceed to secondary school, and Grades 10 and 12 finish with national exams. Grade 10 exams determine if a student continues to preparatory school (Grades 11 and 12), college, or drops out, and Grade 12 exams determine if the student continues to university education. Thus whilst the results of these exams were recorded for purposes of promotion (Grade 10 students who then went to college were considered not promoted), due to their different format they were not included as measures of attainment.

The historic academic data was collected and recorded by academic year, and when tracking a student, it was important to consider all possible outcomes for a student in a given year, in order to be able to find the student in the subsequent year. When a sample student was identified in a given grade and year, whether moving forwards or backwards, it was important to bear in mind that the student could have been promoted to the next grade, have repeated the year, changed school, dropped out, or been promoted by an additional grade (for exceptional attainment), and this was not always immediately clear from the roster. It was also noted that there were a large number of students who would be listed in the roster and therefore registered as enrolled, yet there was little evidence of the student actually attending school in a meaningful way, with no results at the end of either semester. This trend was particularly noticeable in Grade 1 rosters. This trend may help to explain why enrolment rates have increased rapidly in recent years, but this has been matched by high dropout rates.

The timeframe for the data collection was from 1998 – 2007 E.C. (2005/6 – 2014/15 G.C.), as 2007 E.C. was the most recent completed academic year at the time of data collection, and the first Link Ethiopia sponsorship started in 2001 E.C. (2008/9 G.C.), and thus a 10-year timeframe allowed for the possible collection of preintervention data for all Link Ethiopia students. For each student their academic data was recorded up until 2007 E.C., or until they dropped out or appeared to leave the school with an ultimately unknown outcome, and was recorded 'backwards' until their first year attending Grade 1, or to when they first joined the school that was in the sample frame (they may have attended a non-sample school previously), or until 1998 E.C.

Finally, whether a student had dropped out from school was not always a clear cut case. Merely the absence of a student from the roster did not necessarily indicate dropout. According to Link Ethiopia staff, school directors and record keepers, students could often change school, or move away from the area without notifying the school. Thus unless dropout was certainly reported by Link Ethiopia or the school, if a student was missing from the roster, dropout was only recorded if the previous year there was evidence of a 'bad' outcome e.g. the lack or exam results from the first or second semester indicated that student had not attended school throughout the year, or a failure to be promoted to following grade.

The data collection process yielded a large amount of missing data, due to a number of reasons such as: students moving both away from and to the sample schools, a lack of historic information on the school and grade of the student to be able to identify them in school records, name changes of students, missing rosters, or because a student dropped out too long ago to find their academic records or to make contact with the student. As mentioned in the sampling section this meant that for some students no historic academic study was found, and therefore these students were removed from the dataset. In spite of this, over 900 observations for each outcome variable were collected, and the study was not compromised.

4.4. Semi-Structured Interviews

Semi-structured interviews, were carried out with nine sponsored students across the two organisations and were based around two main themes. One set of questions was geared towards trying to understand the relative poverty of the students and their families, to assess how effective the targeting of the programmes was towards those students most in need, and to understand to what extent the sponsorship schemes appropriately addressed the barriers to education associated with poverty. The second main area to explore was the effectiveness of the scheme from the point of view of the students. The students were asked what their experience of being a sponsored student was like, the impact of sponsorship on both them and their family, which parts of the sponsorship they valued more than others, and their recommendations for improvements to the programme.

The interviews also began with an introductory section about the students' experience at school to 'settle' interviewees into the interview, and to understand the educational aspirations of sponsored students. The semi-structured interviews focussed on these issues to complement the quantitative data, and to answer questions that the quantitative data could not. The quantitative data can demonstrate the size of the impact of the sponsorship, but it cannot answer why it is or isn't working, or which aspects of the programme are working and which aspects aren't in as nuanced a way. Furthermore, the lack of quantitative background socio-economic data available for both the treatment and control students meant that the effectiveness of the selection process had to be explored through qualitative methods.

5. Empirical Strategy

5.1. Research Questions

Based on the mission of Link Ethiopia (see Section 3.2.1), the design and intent of the sponsorship programme, and the availability of data, it became clear that two main research questions could be addressed. Firstly, this study ultimately seeks to answer the question '*Are Child Sponsorship Programmes an Effective way to Improve Educational Outcomes?*'; that is can a positive causal relationship between sponsorship programmes improved access to, and the quality of education be identified? Access implies the basic ability to attend school, on the basis that sponsorship reduces the private costs to education, and quality alludes to improved school performance when attending school due to improved access to, and resources at, school. Secondly, by comparing the results between the students sponsored by Link Ethiopia and Kindu, a secondary research question can be addressed that is, '*Do Conditional Cash Transfers, in addition to the provision of School Resources, Improve Educational Outcomes*?'.

5.2. Outcome Variables

Both programmes are relatively young, in that there is no data on the scale required, as to whether students completed primary, secondary or tertiary education, or on the total completed years of schooling, to carry out the analysis on the same outcomes as in the Compassion analysis (Wydick et al. 2013), as the majority of students in both schemes are still in education. The use of educational annual data for each student though, can in many ways be considered a strength of this study.

Whilst there may be possible spillover effects from the sponsorship programme, both within households and within schools, a lack of data on other children in households of sponsored children, or control schools makes this difficult to estimate these. Intrahousehold spillovers would probably be considerably larger for the Kindu model, as the family benefits from both health checks and the CCT, however evidence for intrahousehold spillovers in the Compassion sponsorship programme was negligible, (although the Compassion programme does not include CCTs or family health checks). Likewise, evidence for intra-village spillovers was negligible (Wydick 2013),

and the number of sponsored students in a given school are such a small proportion of total students, no more than 30 in a school of 1000, that intra-school spillovers would be expected to be minimal as well. Thus the failure to account for spillovers should not have a major impact on the results.

5.2.1. Access

To measure the impact on access this paper focuses on two core outcome variables, the first of which is **enrolment rates**. Enrolment rates are estimated based on the binary outcome of either enrolled or not enrolled, for a given student, in each academic year, from the first (chronological) observation until the 2014/15 academic year or the last known observation. This study uses a stricter definition of enrolment than is normally applied. Enrolment normally implies that a student is registered at a school, however in this study if the roster showed no evidence of the student attending school in a given year, this student was considered to be not enrolled. The enrolment variable gives us a clear indication of the number of years that a student attended school.

The second 'access' variable is attendance, which in this case is measured in terms of the discrete variable, '**days absent**'. Attendance is an important measure of access to education, as a student who attends school on half the days compared to another student, say due to family work obligations, has a much lower exposure to education, but this is not captured in the binary enrolment variable.

5.2.2. Quality

The improved quality of education should manifest itself in improved academic performance from the sponsored students, and this can be measured in two ways. The first is a simple measure of attainment, that is the **average** result from internal exams in a given year.

The second set of variables relate to progress through the academic system, given that repetition of grades, or years absent from education are common in the Ethiopian education system. Progress is measured in two ways, firstly, through **promotion rates** (which are estimated in the same way as enrolment rates), and
secondly using ratios comparing expected grade based on age (dividing the actual grade by the expected grade).

Expected Grade, in this study, is calculated in two different ways. '**Grade Ratio 1**' calculates the expected grade based purely on age. In Ethiopia students are expected to start Grade 1 aged 7, and therefore students aged 8 should be in Grade 2 and so on. This ratio gives an overall indication of the educational progress for the student in relation to their age. '**Grade Ratio 2**' calculates the expected grade based on the age and grade of the student at the first (chronological) observation. For example, if the student is first observed in Grade 1 aged 10, then aged 11 they would be expected to be in Grade 2.

For Grade Ratio 1, a student who starts school at a later age, but then progresses efficiently through the system, will have a low grade ratio for the early grades, but that ratio will improve as they progress through the system, reflecting that that student is 'catching up' with their peers, and that the grade they are in is now more reflective of their age than when they started school. Grade Ratio 2 ignores the actual age of the student and only measures how efficiently a student progresses through the education system, and will mean that such a student will consistently have a 'high' grade ratio.

For both grade ratios, the maximum value for each observation is set to 1, in that if a student is in a grade at a younger age than expected, say due to starting school at a younger age, this is not considered to suggest that the student is progressing through the system more efficiently, especially as this study is more concerned with whether students are 'falling behind' in their education to not. If such an approach was not used, then a student aged 7 in Grade 2, would have an expected grade of 1, and thus a grade ratio of 2, and given that most values for the grade ratios are 1 or just below such a value would disproportionately affect the results.

If a student dropped out, then the grade they were in at the time of dropout, is then held fixed at that grade from the year of dropout through to the 2014/15 academic year, and thus the grade ratio for that student would decline over time, and therefore these ratios can be seen as composite indicators of both progress and access.

These two ratios may yield different results, as in Ethiopia many students start Grade 1 older than aged 7. This is illustrated by the fact that the Apparent Intake Rate (number of students enrolled in Grade 1 divided by population aged 7) of 152.9% is much higher than the Net Intake Rate (number of students aged 7 enrolled in Grade 1 divided by population aged 7) of 105.6% in 2013/14 (Ministry of Education 2015b).⁶

5.3. Difference-in-Difference Analysis

Whilst the results section presents some initial descriptive statistics showing the difference between the sponsored and control students, difference-in-difference analysis is used to understand the programme effects with greater precision. Although the preference for prospective analysis was stated, by definition the impact evaluation of this scheme is retrospective, as it requires the analysis of existing data of the programme to date. In this analysis the 'input of interest' [sponsorship] does vary in accordance with unobserved variables, as the sponsorship is means-tested, and the baseline 'poverty scorecard' used in the selection process is not included in the data set, this analysis should be wary of any omitted variable bias, that may upwardly bias the programme effect. As discussed though, in such a case, a difference-in-difference approach can go a long way to reducing the size of this bias.

The difference-in-difference analysis comes close to an experimental approach to research, even when gathering data retrospectively post-treatment, as the historic data available means that both the treatment and control students can be observed both pre- and post-treatment. Importantly, a difference-in-difference approach accounts for the pre-existing differences in the treatment and control groups. In this case, prior to intervention the sponsored students would be expected to have poorer educational outcomes, as the selection of students for sponsorship is means-tested, and access to education is unequal across socio-economic groups (Ministry of Education 2015a: 34).

The data from the control group allows the estimation of what would have happened to the treatment group in the absence of the treatment. A 'parallel-trend assumption'

⁶ Net Intake Rate should theoretically not exceed 100%, these results are probably due to underestimates in population

is applied, as is the norm in difference-in-difference analysis, that in the absence of treatment, the outcomes of the sponsored students would have the same trend (slope) as the control group. The trend of the control group represents the unobserved changes that impact on educational outcomes in Link Ethiopia schools over the time-period. The difference-in-difference approach can be understood as a regression on the outcome of interest. The basic regression model (5.1) is below:

$$Y_{it} = \beta_1 T_i + \beta_2 t + \beta_3 (S_i, t) + \beta_4 (X) + \varepsilon_{ijklt}$$
(5.1)

Where:

 Y_{it} = the educational outcome for a student *i* at time *t*

Where: *i* = 1,...,*N*

Where: N = the finite number of students in the study

And: $t = (t_{1,...,}t_{10})$ is time-series data across 10 evenly spaced and of equal length time periods (academic years). Here t_1 is the academic year 2005/06 and t_{10} is the academic year 2014/15

 $T = \{0,1\}$, a dummy variable determined by whether the student is part of the control group [T=0] or the control group [T=1]

S = the treatment variable that interacts with t.

 $\beta_{1,\ldots,4}$ = constants to be found in the regression model.

 β_3 = the constant of interest as this is the coefficient of the programme effect.

X = a vector of control variables, which may affect the outcome in question.

 ε_{ijklt} = is the error term where:

j = school index

k = grade index

I = age in years (discrete values)

With there being multiple time periods the model is more complex than a simple difference-in-difference model, in that the time trend observed is not as simple as pre- and post-intervention, but rather there are multiple observations both pre- and post-intervention.

The nature of the treatment variable *S* is a major methodological consideration, and this study employs four different models of *S*. In many difference-in-difference analyses $S = \{0,1\}$, where 0 is the value applied for all observations before the student receives treatment (including never receiving treatment), and 1 is applied to all observations that occur post-intervention (irrespective of whether the treatment was received in that year or not). This can be considered the basic model 'Model 1'.

Alternatively, we can view the treatment (or non-treatment) of sponsorship as something that occurs on an annual basis, where *S* is a continuous variable representing the number of months that a student received sponsorship in a given academic year. This 'Model 2' may be considered superior to Model 1, in that it differentiates between those students that have received sponsorship and continue to receive sponsorship, and those students where sponsorship has been received but has now been terminated.

In the Link Ethiopia programme sponsorship can be terminated for one of two reasons. The first is that the family situation of the student improves, such that they are considered to no longer need sponsorship. Much more common though, is that the sponsorship is terminated due to the child having dropped out of school, with no intent to return to education. This model is therefore more effective in estimating the impact of the programme in terms of the expenditure of resources in the programme. It may be considered, however, that by not including the observations of dropped out students in the treatment variable, as once the student has dropped out S = 0, the size of the programme effect may be upwardly biased. It should be noted that when a student has dropped out, the data for promotion, days absent, and average is not recorded, and thus the results from these outcomes are not affected by dropped out students. The results for the outcomes of enrolment and grade ratios will, however, be affected.

Alternatively, it is possible that the impact of sponsorship may be linked to the length of sponsorship, in that a student who had received two years of sponsorship would be expected to have better educational outcomes than a student who had received sponsorship for one year. This logic is applied in 'Model 3', where *S* is the cumulative months of sponsorship at time *t*, and in this model, for dropped out students who are no longer sponsored, the treatment variable remains as the total months of sponsorship from when dropout occurs through to t_{10} .

In 'Model 4', a non-linear relationship between the cumulative months of sponsorship and the outcome variables is investigated, where a positive coefficient would indicate that the benefits to sponsorship accrue exponentially with time sponsored. In such a case, if the turning (minimum) point on the regression line is where (S > 0) then the effect of sponsorship is below the parallel trend up until the turning point, at which point the effect will then then tend back towards 0 and the continue to increase exponentially.

For the remainder of this paper *S* can be any of S_1 , S_2 , S_3 or S_4 where:

 $S_1 = \{0,1\}$, a dummy variable determined by whether the observation at time *t* occurs before student receives or if a student never receives sponsorship [$S_1=0$] or after receiving sponsorship [$S_1=1$]

 S_2 = the number of months that sponsorship is received during the time period *t*.

 S_3 = the cumulative number of months of sponsorship has been received at time *t*.

 $S_4 = S_3 + (S_3)^2$

5.3.1. Control Variables

Control variables are introduced into the model, to account for any difference in the outcomes that is due to differences in observed characteristics of the students in both the treatment and control groups, and to increase the explanatory power of the model in relation to its ability to predict the outcome variable, and reduce standard errors. Of the data available there are four control variables that warrant inclusion in the model.

The first is **gender**. As mentioned previously, in Ethiopian primary school girls perform as well as or better than boys, but do not perform as well as boys in secondary school (Gretland et al. 2014; Ministry of Education 2015a).

The **school** of a given student is controlled for to account for school quality, which may vary due to differences in resources, administration quality and school-level training.

Grade is controlled for, as dropout rates vary from grade to grade. For example, the dropout rate is 21.8% in Grade 1, 11.6% in Grade 5, and 10% in Grade 8, but as low as 1.5% in Grade 7, 3.1% in Grade 6 and 5.4% in Grade 3 (Ministry of Education 2015b). It is also reasonable to expect that exams may vary in difficulty from grade to grade, affecting attainment and progress.

Finally, **age** is controlled for, as within a given grade, an older student may be expected to perform better, especially as this may be their second or third attempt at the grade. Given that both age and grade in this dataset are discrete data and therefore dummy variables, the addition of controls to equation 5.1 produces equation 5.2:

$$Y_{it} = \beta_1 T_i + \beta_2 t + \beta_3 (S_i.t) + \beta_4 Gender_i + \sum_{j \in J} \beta_j [School_{it} = j]$$

$$+ \sum_{k \in K} \beta_k [Grade_{it} = k] + \sum_{l \in L} \beta_l [Age_{it} = l] + \varepsilon_{ijklt}$$
(5.2)

This regression is applied to all outcome variables, except when the outcome is a grade ratio, where grade is removed as a control variable, as it is not independent to the outcome variable in this case.

5.3.2. Fixed Effects

For each treatment model fixed effects are applied independently at two different levels to implicitly control for everything that is fixed about a group of observations, even when those characteristics are not observed in the data set. Fixed effects are used when the outcomes within a group of observations are expected to be correlated due shared characteristics.

Thus, in equation 5.3 fixed effects are applied at a student level (α_i) , to account for anything that is unobserved for the student, such as socio-economic status or IQ. In equation 5.4 observations are categorised into groups by school and grade and fixed effects are applied at this level (α_{ik}). This fixed effects accounts for differences in school and grade, for the same reasons as to they are included as control variables. As some students attend more than one school through the time observed, these fixed effects are applied independently rather than in the same model, as the student-level observations are not always nested within the "school-grade" groups. For both equations 5.3 and 5.4, standard errors clustered at the same level as fixed effects are applied, as the standard errors within these groups are also expected to be correlated.

$$Y_{it} = \beta_1 T_i + \beta_2 t + \beta_3 (S_i.t) + \beta_4 Gender_i + \sum_{j \in J} \beta_j [School_{it} = j]$$

$$+ \sum_{k \in K} \beta_k [Grade_{it} = k] + \sum_{l \in L} \beta_l [Age_{it} = l] + \alpha_i + \varepsilon_{ijklt}$$

$$Y_{it} = \beta_1 T_i + \beta_2 t + \beta_3 (S_i.t) + \beta_4 Gender_i + \sum_{l \in J} \beta_j [School_{it} = j]$$
(5.3)

$$+\sum_{k\in K}\beta_{k}[Grade_{it}=k] + \sum_{l\in L}\beta_{l}[Age_{it}=l] + \alpha_{jk} + \varepsilon_{ijklt}$$
(5.4)

i∈I

Thus, ultimately there are eight possible models that estimate the programme effect, as there are two different models (equations 5.3 and 5.4) depending on the level at which fixed effects are applied, and four different models of the treatment effect S. These eight models can be tested across all outcome variables, with the exception of grade ratios as school-grade fixed effects cannot be applied to the grade ratio outcomes, as fixed effects need to be independent of the outcome variable in the same way that control variables do.

Finally, it is worth noting that certain variables 'drop out' or are 'absorbed' by the fixed-effects parameter, when the variable is fixed across all observations for that group. When student fixed effects are applied as in equation 5.3, the treatment and gender variables are absorbed resulting in equation 5.5. When the school-grade fixed effects are applied as in equation 5.4, the school and grade variables are absorbed resulting in equation 5.6.

$$Y_{it} = \beta_2 t + \beta_3 (S_i.t) + \sum_{j \in J} \beta_j [School_{it} = j] + \sum_{k \in K} \beta_k [Grade_{it} = k]$$

$$+ \sum_{l \in L} \beta_l [Age_{it} = l] + \alpha_i + \varepsilon_{ijklt}$$

$$Y_{it} = \beta_1 T_i + \beta_2 t + \beta_3 (S_i.t) + \beta_4 Gender_i + \sum_{l \in L} \beta_l [Age_{it} = l]$$

$$+ \alpha_{jk} + \varepsilon_{ijklt}$$
(5.5)
(5.6)

5.3.3. Weights

A simple weighting is applied to the regression such that each student is equally weighted in the analysis. Before the application of weights there are different numbers of observations for different students, thus biasing the results towards the outcomes of students for which there are more observations. By weighting in this way the effect estimated is the average treatment effect per student.

5.3.4. Comparing Kindu and Link

As mentioned in the sampling section, the sample of beneficiaries sponsored by the Kindu Trust was drawn from a database of students currently in education. This meant that when carrying out analysis comparing the Link Ethiopia students, the Kindu students, and the non-sponsored students, all students that had dropped out from education were removed from the datasets to allow for unbiased comparisons between the groups. For this reason, the outcome variable of enrolment became obsolete, and so is not explored when comparing Link and Kindu students, however

all other outcomes were still tested, using the same difference-in-difference approach, and same eight models outlined above.

Given this new data set, the first step was to run the difference-in-difference analysis, again comparing the Link and non-sponsored students, to establish if the impact of sponsorship was similar, even with the new dataset. The next step was to then compare the Kindu students to the non-sponsored students in the same way, to look for evidence of the impact of the Kindu sponsorship programme. Then finally, difference-in-difference analysis was run comparing the Kindu students to the Link Ethiopia students, treating the Link Ethiopia students as the control group (as both groups receive school resources), and Kindu students as the treatment group, as they are the group to receive the CCT in addition to school resources.

6. Results and Interpretation

This section is divided into two parts, the first section addresses the first research question, 'Are Child Sponsorship Programmes an Effective way to Improve Educational Outcomes?', by comparing the outcomes of Link Ethiopia students against the non-sponsored students. The second section seeks to answer the question, 'Do Conditional Cash Transfers, in Addition to the Provision of School Resources, Improve Educational Outcomes?'. This section uses a second data set, comparing the outcomes of Link, Kindu and non-sponsored students, all of whom were, in their most recent observation, in education.

6.1. Are Child Sponsorship Programmes an Effective way to Improve Educational Outcomes?

This section starts with some "naïve" descriptive statistics comparing the key outcome variables for the Link Ethiopia sponsored students with the non-sponsored students, without any adjustments such as weights or otherwise. The results of the difference-in-difference analysis comparing the two groups follow the descriptive statistics.

6.1.1. Descriptive Statistics

Table 3 shows the outcomes of interest for the two groups of students, those not sponsored, and those sponsored by Link Ethiopia, as either percentages (for binary variables), or as means (for continuous variables), or in the case of days absent, the estimated mean. These initial results would seem to suggest an effective programme, with Link students exhibiting higher enrolment and promotion rates, missing less days of school, and progressing faster through the education system.

Sponsorship Status	Enrolment Rate	Days Absent	Promotion Rate	Grade Ratio 1	Grade Ratio 2	Average
Not Sponsored	80.11%	14.11	83.87%	0.74	0.88	66.24
Sponsored by Link	93.37%	9.85	90.06%	0.86	0.95	64.98
Total	85.82%	12.05	86.77%	0.79	0.91	65.65

Table 3: Comparing Link Ethiopia and Non-Sponsored Students

The only outcome for which the Link Ethiopia students do not perform as well as the non-sponsored students is in pure attainment terms, with their average results being slightly lower than the non-sponsored students.

For the non-binary variables one can look beyond the means and into the distribution of observations for each outcome. In the following boxplots group 0 are the non-sponsored students, and group 1 are the Link Ethiopia students. Figure 5 reinforces the suggestion that sponsored students miss less days of school than their non-sponsored counterparts, with lower values for each quartile for sponsored students. Likewise, when looking at the grade ratios in Figure 6, Q_1 and Q_2 have higher values for Link Ethiopia students than non-sponsored students, suggesting that for their age sponsored students have generally progressed further through the education system. In Figure 7, Q_1 , Q_2 and Q_3 are all equal to 1 for sponsored students, but for non-sponsored students, whilst Q_3 and Q_2 are equal to 1, Q_1 is lower for non-sponsored students than Link students, reflecting the higher promotion rates for Link students. For both groups across both grade ratios a lot of observations are equal to 1, partly due to 1 being the maximum value, but also due to 1 being the value representing the expected rate of progress through the education system.

When looking at attainment in Figure 8, Q_1 and Q_2 are very similar across both groups, however Q_3 is higher for non-sponsored students than Link students, suggesting that the top half of students in this group outperform the top half of Link Ethiopia students, whereas the bottom halves of each group perform similarly.

These results suggest that sponsored students have better access to education, and progress faster through the education system, although they perform slightly worse in attainment terms. To establish a causal relationship between sponsorship and improved outcomes though, requires the difference-in-difference analysis.



Figure 5







Figure 7



Figure 8

6.1.2. Difference-in-difference analysis

For each of the outcome variables, difference-in-difference analysis was carried out, as outlined in the empirical strategy. As mentioned there were four different models of treatment effect, and for each model, fixed effects either at a student, or a school-grade level were applied independently. The table below gives a summary of the results for each outcome variable, where yes means the programme effect was significant to a 95% confidence interval. Of the significant results, all were positive programme coefficients.

	Model	1 ¹	Model 2 ²		Model 3 ³		Model 4 ⁴	
Fixed Effects	Student	SG⁵	Student	SG⁵	Student	SG⁵	Student	SG⁵
Enrolment	No	No	No	No	No	No	Yes	Yes
Days Absent	No	No	No	No	No	No	No	No
Promotion	No	Yes	No	No	No	No	Yes	Yes
Grade Ratio 1	Yes	N/A	Yes	N/A	No	N/A	No	N/A
Grade Ratio 2	Yes	N/A	Yes	N/A	Yes	N/A	No	N/A
Average	No	No	No	No	No	No	No	Yes

Table 4: Summary of Difference in Difference Results: Link vs. Non-sponsored students

1: All observations post-treatment

2: Months of sponsorship student received in a year

3: Cumulative months of sponsorship for student

4: Cumulative months of sponsorship + (Cumulative months of sponsorship)²

5: School-Grade fixed effects

Yes = p < 0.05

As Table 4 shows, all significant results are found, with the exception of the grade ratios, and one result for promotion for Model 1 with school-grade fixed effects, for Model 4, suggesting that for most educational outcomes there is a non-linear relationship between sponsorship duration and educational outcomes. There are different possible explanations of this, but the implication is that the benefits to sponsorship increase with the duration of sponsorship, rather than sponsorship being merely a repeated annual treatment, and that this impact increases exponentially; the longer the duration of sponsorship the greater the benefits of an

additional month of sponsorship. For all of these significant results the turning (minimum) points of the regression line occur when S > 0, indicating that in the first years, the average effect per student is below the parallel trend, but after this turning point (as the duration of sponsorship increases) the average effect per student tends back towards, and then exceeds 0.

The possible reasons for this relationship between sponsorship and outcome variables include self-selection, where sponsorship is ineffective for a subset of students who dropout with in the first years of sponsorship; time lags, where the sponsorship takes times to have an impact; the 2012/13 increase in individual support for students; and also a possible failure of difference-in-difference analysis to fully capture the impact of sponsorship. How these factors would contribute to this non-linear relationship is explored further in the 'Discussion'.

The evidence for a relationship though, between duration of sponsorship and educational outcomes is very convincing. Therefore, whilst significant results for grade ratios for Models 1 and 2 are found, Model 3 is the preferred model as this corresponds with the relationship between length of sponsorship and educational outcomes. Grade ratios may not follow Model 4 as the way they have been constructed, comparing expected and actual grades and thus incorporating previous historic outcomes in each observation, and having a maximum value of 1, would not lend itself to a non-linear relationship.

In the following regression tables only the coefficients of the treatment variables are shown, as the large number of categorical variables used as control variables means that including the full regression tables is not practical.

Access

The outcomes of interest in relation to access are enrolment, and the number of days absent. As noted above, the significant results for enrolment are for Model 4, and thus Table 5 shows the results for this model. It shows that the non-linear treatment model is robust to both student and school-grade level fixed effects at a 95% confidence level, suggesting a strong result.

Enrolment Rate	Student Fixed Effects	School-Grade Fixed Effects
e	-0.00356*	-0.00265
53	(0.017)	(0.107)
$(\mathbf{C})^2$	0.0000499*	0.0000424*
(33)	(0.029)	(0.044)
Constant	1.216	1.276***
Constant	(1.000)	(0.000)
Observations	1219	1219

Table 5

p-values in parentheses

 $p^* < 0.05, p^* < 0.01, p^* < 0.001$

In the case of the enrolment variable, the self-selection hypothesis would seem the more plausible, as after a student drops out, in subsequent years they were recorded as not enrolled, and the cumulative months of sponsorship would have remained fixed for these students, but for those who remain in the sponsorship scheme they continue to be enrolled in school as the duration of sponsorship increases.

For the student level fixed-effects model for the first 35.67 months of sponsorship the average effect per student is negative and declining, and thereafter the marginal effect per student of an additional month of sponsorship is positive, and it takes 71.34 months for the net average effect of the sponsorship on enrolment to be positive, and from there on the average effect continues to rise. In the school-grade fixed effects the turning point is 31.25 months, taking 62.50 months for the average effect to become positive. Whilst the case for sponsorship impacting enrolment is strong, there was however, no evidence of a statistically significant relationship between the sponsorship at the number of days absent from school.

Progress

The most rudimentary measure of progress through the education system, is whether a student is promoted to the next grade at the end of each year. And as with enrolment, Model 4, indicating a non-linear relationship between promotion rates and sponsorship duration yielded a statistically significant result.

Promotion Rate	Student Fixed Effects	School-Grade Fixed Effects
6	-0.00428	-0.00524*
33	(0.127)	(0.028)
$(0)^2$	0.0000875*	0.000103**
(33)	(0.049)	(0.002)
Constant	0.385	0.514
Constant	(0.283)	(0.053)
Observations	933	933

Table 6

p-values in parentheses

p < 0.05, *p* < 0.01, *p* < 0.001

Table 6 shows that for Model 4 the result is again statistically significant when either student or school-grade fixed effects are applied. In the student fixed effects model, the turning point is slightly earlier than in the enrolment model, at 24.46 months, with sponsorship having a net positive effect after 48.91 months, and in the school-grade fixed effects model, the turning point is at 25.44 months, with a net positive effect after 50.88 months. This earlier turning point than in the enrolment example may be due to promotion data not being recorded once the student has dropped out (as evidenced by the lower number of observations). Thus these results add weight to the self-selection interpretation of the data, as grade repetition and dropout are linked (Ministry of Education 2015a), and therefore these students may be causing the negative average effect on educational outcome in the first years of sponsorship.

When looking at ratios of actual grade to expected grade, significant results for both ratios are found not for Model 4, but for both Models 1 and Model 2 with fixed-effects at a student level (Table 7 and Table 8). The positive coefficients in these models demonstrate that after sponsorship, and whilst continuing to receive sponsorship that students are more likely to be in a higher grade than if they were not sponsored. As mentioned earlier, a positive effect on Grade Ratio 1 implies a faster catching up in grade related to age, and a positive effect on Grade Ratio 2 implies faster progression through the education system. For Grade Ratio 2 there is also a

significant result for the preferred Model 3, which indicates a relationship between the duration of sponsorship and a student's overall progression through the education system.

Grade Ratio 1	Model 1	Model 2	Model 3
c	0.0393*		
5 1	(0.022)		
c		0.00371*	
32		(0.021)	
c			-0.000371
33			(0.457)
Constant	0.810***	0.810***	0.802***
Constant	(0.000)	(0.000)	(0.000)
Observations	1189	1189	1189

Table 7

p-values in parentheses p < 0.05, p < 0.01, p < 0.001

Grade Ratio 2	Model 1	Model 2	Model 3
e	0.0613***		
31	(0.000)		
c		0.00645***	
32		(0.000)	
c			0.00188***
33			(0.000)
Constant	1.321***	1.324***	1.292***
Constant	(0.000)	(0.000)	(0.000)
Observations	1218	1218	1218

Table 8

p-values in parentheses p < 0.05, p < 0.01, p < 0.001

As mentioned in the empirical strategy, these grade ratios are composite indicators of both enrolment and promotion. These results do suggest that sponsored students have currently progressed further through the education than they would have done in the absence of sponsorship. Given the combined evidence from both the grade ratios, and the promotion rates, there is strong evidence that the sponsorship programme has a positive effect on educational progress, however again, the positive impact may be limited to a subset of students.

Attainment

The only result to yield a significant result on the outcome of average score is for Model 4, when school-grade fixed effects are applied. No significant results are found when student fixed effects are applied, and whilst the result in this model is nowhere near statistical significance, it is worth noting that it produced a negative rather than a positive effect. Given that there is only a significant effect when schoolgrade fixed effects are applied the evidence for sponsorship having an effect on average score is not particularly robust.

Average	Student Fixed Effects	School-Grade Fixed Effects		
6	0.0385	-0.105		
S 3	(0.575)	(0.225)		
$(\mathbf{C})^2$	-0.000865	0.00219*		
(53)	(0.365)	(0.042)		
Constant	37.87***	61.86***		
Constant	(0.000)	(0.000)		
Observations	1017	1017		

Table 9

p-values in parentheses

p < 0.05, p < 0.01, p < 0.001

6.1.3. Impact of Link Ethiopia Sponsorship

There is strong evidence that the sponsorship scheme can have a positive impact on educational outcomes for students, particularly in relation to access to education and progress through the educational system. Improvement in educational outcomes is though reliant on the student remaining in the sponsorship programme for a sufficient period of time. Typically, to see an improvement in both enrolment and promotion rates the sponsorship has to last longer than a little under 6 years. The evidence for a positive impact on the attainment of sponsored students is less convincing, however given that promotion is contingent on reaching a minimum level of attainment, there can be considered to be a link between sponsorship and meeting this required level.

The strong results for a positive impact on grade ratios suggest that on average sponsored students progress further through the education system that in the absence of sponsorship. The collective evidence though, notably the non-linear relationships between cumulative months of sponsorship and both enrolment and promotion, seems to suggest that the sponsorship is only effective for a subset of students, and that for the other students, the sponsorship is not effective. The evidence from these results would suggest that this non-linear relationship is due to self-selection within the data, and that some students do not benefit from sponsorship, whilst others do, however competing explanations for this non-linear relationship are explored further in the 'Discussion'.

6.2. Do Conditional Cash Transfers, in Addition to the Provision of School Resources, Improve Educational Outcomes?

This analysis is again divided into descriptive statistics and the difference-indifference analysis. Using the data set of students who were still in education at their most recent observation, this analysis compares three groups, Link Ethiopia students, Kindu students and non-sponsored students.

6.2.1. Descriptive Statistics

Table 10 shows the key outcome variables for the three groups of students from the dataset. As expected for such students, the enrolment rates are very high across all

groups, with only the occasional student missing a year of education and then returning to school, and thus do not warrant further investigation. The results initially seem similar to the original comparison between Link Ethiopia and non-sponsored students, with both Kindu and Link students missing less days of school, whilst having slightly lower levels of attainment than the non-sponsored students. Likewise, for Grade Ratio 1 both Kindu and Link groups have a higher mean than the nonsponsored students, suggesting that these students have progressed further through the education system in relation to their age. Unlike the original observations though, the promotion rates and means for Grade Ratio 2, are similar across all groups, whereas the original analysis comparing Link and non-sponsored students showed better results for these variables for sponsored students compared to non-sponsored students.

Sponsorship Status	Enrolment Rate	Days Absent	Promotion Rate	Grade Ratio 1	Grade Ratio 2	Average
Not Sponsored	99.60%	10.37	89.80%	0.81	0.97	67.27
Sponsored by Link	98.67%	8.37	91.67%	0.88	0.96	65.45
Sponsored by Kindu	98.26%	5.05	90.27%	0.93	0.98	64.52
Total	98.98%	8.64	90.60%	0.86	0.97	66.04

 Table 10: Comparing non-sponsored students with both Kindu and Link Ethiopia sponsored students

Again, for the non-binary variables boxplots can reveal greater insight into the distribution of outcomes for each group. In the following boxplots group 1 are the non-sponsored students, group 2 are the Link Ethiopia students and group 3 are the Kindu students. The boxplot for Figure 9 illustrates that for days absent there are similar skews in each group, with the quartiles having higher values for sponsored students, perhaps suggesting that the CCT boosts school attendance.

Figure 10 clearly suggests that sponsored students tend to progress faster through the education system in relation to their age than non-sponsored students, and that Kindu students progress faster than Link students. For the Kindu group Q_1 , Q_2 and Q_3 are all equal to 1, and for Link students Q_2 and Q_3 are equal to 1, and for the nonsponsored group only Q_3 is equal to 1. As Figure 11 shows though, for Grade Ratio 2 there is little evidence of this trend, and that the majority of students progress at the expected rate across all groups and therefore for all groups Q_1 , Q_2 and Q_3 are all equal to 1.

In relation to attainment, Figure 12 shows the distribution of average results, where non-sponsored students slightly out-perform Link Ethiopia students, who slightly outperform Kindu students. It is worth noting that the Kindu students have a much smaller range and interquartile range, however any explanation for this is unclear. Whilst these small differences do exist, they are only small and no conclusions should be drawn at this stage.







Figure 10







Figure 12

6.2.2. Difference-in-difference analysis

The difference-in-difference analysis involves three steps. First, Link Ethiopia students are compared with the sponsored students to see if there are any differences in the results when compared to the original difference-in-difference analysis. Any results should not be considered to either contradict or reinforce the results from the initial analysis, however some of the results do suggest that self-selection may not be the sole cause of the non-linear relationship, and this should not be ignored.

This comparison is primarily intended to help to understand the effect of limiting the dataset to students who are still enrolled. This enables better understanding of the results when comparing the Kindu students to the non-sponsored students when looking for evidence of the effectiveness of the Kindu programme. For example, for a given variable there may have been an effect when comparing the original Link and non-sponsored students, but no effect when comparing the Kindu students with the non-sponsored students in the second dataset. If when comparing the Link and non-

sponsored students in the second dataset there is also no effect, then the Kindu results should be understood in this context. Finally, the results between the Kindu students and Link Ethiopia students are compared to see if the conditional transfers, on top of school resources, have a significant impact.

Again, in the following regression tables only the coefficients of the treatment variables are shown, as the large number of categorical variables used as control variables means that including the full regression tables is not practical.

Link vs. Not Sponsored

Table 11 shows the results of the difference-in-difference analysis comparing Link Ethiopia students against non-sponsored students from the second data set. As before there is no effect on attendance, and there is a positive effect on promotion when applying Model 4.

	Model 1 ¹		Model 2 ²		Model 3 ³		Model 4 ⁴	
Fixed Effects	Student	SG⁵	Student	SG⁵	Student	SG⁵	Student	SG⁵
Days Absent	No	No	No	No	No	No	No	No
Promotion	No	No	No	No	No	No	Yes	Yes
Grade Ratio 1	No	N/A	No	N/A	Yes (-)	N/A	No	N/A
Grade Ratio 2	No	N/A	No	N/A	No	N/A	No	N/A
Average	No	No	No	No	No	No	No	No

Table 11

1: All observations post-treatment

2: Months of sponsorship student received in a year

3: Cumulative months of sponsorship for student

4: Cumulative months of sponsorship + (Cumulative months of sponsorship)^2

5: School-Grade fixed effects

Yes = p < 0.05

Table 12 shows the results from the difference-in-difference analysis on promotion between the two groups. This suggests that this relationship may not be the product of self-selection, but that promotion rates across all students decline before improving as the duration of sponsorship increases. Possible explanations for this are found in the 'Discussion'. The time-period for this trend is also the similar to the initial result, with the average effect decreasing for 25.9 months, and taking 51.8 months to return to before-sponsorship levels.

Promotion Rate	Student Fixed Effects	School-Grade Fixed Effects		
<u> </u>	-0.00428	-0.00524 [*]		
5 3	(0.127)	(0.028)		
(0.) ²	0.0000875*	0.000103**		
(53)	(0.049)	(0.002)		
Constant	0.385	0.514		
Constant	(0.283)	(0.053)		
Observations	933	933		

Table 12

p-values in parentheses

p < 0.05, *p* < 0.01, *p* < 0.001

When looking at grade ratios there are also different results to the original analysis. The evidence of positive impact of sponsorship on grade ratios from the moment after sponsorship (Model 1), or in the years when sponsorship is received (Model 2), and the only significant effect is effect on Grade Ratio 1 for Model 3, which is in fact has a negative coefficient. Thus failure of the Kindu programme to improve grade ratios does not suggest that the programme is ineffective.

When looking at the effect of sponsorship on average score, the results from Model 4, shown in Table 13 are similar to the results in the original analysis, however the p-value for the school-grade fixed effects model is slightly higher, meaning there is not a significant result at the 95% confidence interval any more. This may be due to the smaller sample size, rather than a marked change in the nature of the effect, and so little should be read into this.

Average	Student Fixed Effects	School-Grade Fixed Effects
6	0.0691	-0.0961
3 3	(0.337)	d Effects School-Grade Fixed Effects 1 -0.0961 7) (0.293) 24 0.00206 2) (0.070) 9 60.86*** (0.000)
(C) ²	-0.00124	0.00206
(53)	(0.202)	(0.070)
Constant	45.39	60.86***
Constant	(.)	(0.000)
Observations	909	909

Table 13

p-values in parentheses p < 0.05, p < 0.01, p < 0.001

Kindu vs. Not Sponsored

The results for the Kindu sponsored students in relation to the non-sponsored students, shown in Table 14, are largely similar to the results comparing Link and non-sponsored students from the same dataset, however there are some variations.

	Model 1 ¹		Model 2 ²		Model 3 ³		Model 4 ⁴	
Fixed Effects	Student	SG⁵	Student	SG⁵	Student	SG⁵	Student	SG⁵
Days Absent	No	No	No	No	Yes	No	No	No
Promotion	No	No	No	No	No	No	No	Yes
Grade Ratio 1	No	N/A	No	N/A	Yes (-)	N/A	No	N/A
Grade Ratio 2	No	N/A	No	N/A	No	N/A	Yes	N/A
Average	No	No	No	No	Yes (-)	No	No	No

Table 14

1: All observations post-treatment

2: Months of sponsorship student received in a year

3: Cumulative months of sponsorship for student

4: Cumulative months of sponsorship + (Cumulative months of sponsorship)²

5: School-Grade fixed effects

Yes = p < 0.05

As the descriptive statistics indicated, the Kindu students did have better attendance rates, and the difference-in-difference analysis suggests that this may well be due to impact of sponsorship. Table 15 shows that the positive impact on attendance is significant when using Model 3, suggesting that improved attendance increases with the length of sponsorship.

Days Absent	Student Fixed Effects	School-Grade Fixed Effects		
<u> </u>	0.223*	0.0454		
53	(0.049)	(0.257)		
Constant	75.67	-7.867		
Constant	(.)	(0.528)		
Observations	297	297		

Table 15

p-values in parentheses

p < 0.05, p < 0.01, p < 0.001

For attainment, whilst there were no significant results for the average result when comparing Link and non-sponsored students, for Kindu students there was a significant result for the Model 3, with a negative coefficient, the results for which can be seen in Table 16. When looking at both days absent and average, it should be noted in both of these cases the result was only significant in the student fixed-effects model, and not in the school-grade fixed effects, so wariness of the certainty of this result is warranted.

When looking at promotion rates there is some evidence of an effect when using Model 4. The significant result using for Model 4 does hold for school-grade fixed effects, however it does not hold for student fixed-effects, as shown in Table 17, meaning that it is less certain whether the Kindu model is as effective in increasing promotion rates as the Link Ethiopia model.

Average	Student Fixed Effects	School-Grade Fixed Effects		
C	-0.102 [*]	-0.00559		
33	(0.024)	(0.760)		
Constant	19.14	51.78***		
Constant	(.)	(0.000)		
Observations	703	703		

Table 16

p-values in parentheses p < 0.05, p < 0.01, p < 0.001

Promotion Rate	Student Fixed Effects	School-Grade Fixed Effects	
6	-0.00189	-0.00393 [*]	
33	(0.682)	(0.011)	
(C) ²	0.0000799	0.0000250*	
(53)	(0.702)	(0.033)	
Constant	-0.748	0.771***	
Constant	(.)	(0.000)	
Observations	715	715	

Table 17

p-values in parentheses

* p < 0.05, ** p < 0.01, *** p < 0.001

When looking at grade ratios there was only a significant result for Model 3 for Grade Ratio 1, and Model 4 for Grade Ratio 2. As with the analysis of Link students in the second dataset, for Grade Ratio 1 this produced a negative coefficient, and thus it can be dismissed as non-pertinent. For Grade Ratio 2 there is a significant positive effect for Model 4. These are seemingly contradictory results, so the effect of Kindu sponsorship on grade ratios is unclear.

	Grade Ratio 1		Grade Ratio 2	
	Model 3	Model 4	Model 3	Model 4
6	-0.00133**	-0.000598	0.000182	0.00165*
53	(0.002)	(0.431)	(0.614)	(0.047)
(S ₃) ²		-0.00000379		-0.00000770*
		(0.294)		(0.035)
Constant	0.770***	0.766***	1.130	1.160
	(0.000)	(0.000)	(.)	(1.000)
Observations	701	701	723	723

Table 18

p-values in parentheses

p < 0.05, *p* < 0.01, *p* < 0.001

By comparing the Kindu students with the non-sponsored students there is evidence that the Kindu sponsorship programme has a positive impact on attendance and promotion rates, although the evidence for a positive impact on promotion rates is less convincing for Kindu than for Link. The results also suggest some evidence that Kindu students perform below the parallel trend in terms of attainment and the evidence for any impact on grade ratios is unclear. Thus the evidence for the Kindu programme having a positive effect of educational outcomes is limited. This may be due to limited dataset involving only students who were enrolled at the last observation. The scheme is primarily aimed at improving access to education, and thus not being able to measure the impact on enrolment and the subsequent effect on grade ratios may hide the effect of the programme.

Kindu vs. Link Ethiopia

When running a difference-in-difference analysis of Kindu students against Link Ethiopia students, with Link Ethiopia students as a control group, across all outcomes, difference-in-difference analysis yielded no statistically significant results with positive coefficients, suggesting that there was little evidence that CCTs, in addition to the provision of school resources improve educational outcomes.

Following this, the difference-in-difference analysis was reversed, treating Kindu students as the control group and Link students as the treatment group, the results of which can be seen in Table 19. Whilst with exception of CCTs they are largely similar programmes, in the provision of school resources, there could be differences in efficiency in the administration of the programme that lead to different outcomes for the two groups. The Link Ethiopia scheme also includes greater school resources in terms of school bags and shoes, and this could also result in better outcomes for Link Ethiopia students compared to Kindu students.

	Model	1 ¹	Model	2 ²	Model	3 ³	Model	4 ⁴
Fixed Effects	Student	SG⁵	Student	SG⁵	Student	SG⁵	Student	SG⁵
Days Absent	No	No	No	No	No	No	No	No
Promotion	No	No	No	No	No	No	No	Yes
Grade Ratio 1	Yes	N/A	Yes	N/A	No	N/A	No	N/A
Grade Ratio 2	No	N/A	No	N/A	No	N/A	No	N/A
Average	No	No	Yes	No	No	No	No	Yes

Table 19

1: All observations post-treatment

2: Months of sponsorship student received in a year

3: Cumulative months of sponsorship for student

4: Cumulative months of sponsorship + (Cumulative months of sponsorship)²

5: School-Grade fixed effects

Yes = p < 0.05

The table suggests that there is no evidence that the Link Ethiopia scheme is any more effective than the Kindu scheme in terms of improving attendance or Grade Ratio 2. As Table 20 shows though, when looking at Grade Ratio 1, these values improve more for the Link students than for the Kindu students, under models 1 and 2, suggesting the Link programme is better at supporting older students 'catch-up' in their education.

The results may also be in part due to positive effect of sponsorship on promotion in Model 4. Given that this result only holds for school-grade fixed effects, and not-

student fixed effects, see Table 21, and that a theoretical explanation for such a nonlinear relationship is unclear, firm conclusions cannot be drawn from this.

Grade Ratio 1	Model 1	Model 2
C	0.0323*	
51	(0.033)	
0		0.000952**
52		(0.009)
Constant	0.884	0.917
Constant	(.)	(.)
Observations	646	646

Table 20

p-values in parentheses p < 0.05, p < 0.01, p < 0.001

	Мо	del 2	Мо	del 4	
Average	Student Fixed Effects	School-Grade Fixed Effects	Student Fixed Effects	School-Grade Fixed Effects	
c	0.106**	0.0124			
52	(0.003)	(0.473)			
e.			0.101	-0.225 [*]	
J 3			(0.214)	(0.022)	
(S ₃) ²			-0.00173	0.00359**	
			(0.097)	(0.004)	
Constant	33.04***	54.42***	30.70***	53.63***	
	(0.000)	(0.000)	(0.000)	(0.000)	
Observations	650	650	650	650	

Table 21

p-values in parentheses p < 0.05, p < 0.01, p < 0.001

Finally, when comparing attainment between the two groups there is some evidence of the Link Ethiopia scheme being more effective than the Kindu programme. There are statistically significant results, both for Model 2 and Model 4, the former holding for student fixed effects, but not school-grade effects, and the opposite being true for Model 4. Again, there is little theoretical backing for there being a non-linear relationship, and with the months of sponsorship model only holding for student-fixed effects, the evidence for an impact on attainment is ultimately limited.

6.2.3. Comparing the two sponsorship programmes

Overall, there is some suggestion that the Link Ethiopia programme, may be more effective than the Kindu programme in improving educational outcomes, in relation to progress and attainment. Ultimately though, the evidence-base is not strong enough to draw solid conclusions from. There was no evidence for the Kindu programme being more effective than the Link Ethiopia programme, which is surprising given that it has the additional CCT. It may be that CCTs may be an effective way to boost access to education, but not necessarily progress and attainment, however due to the limited data employed, such a theory has not been able to be tested.

7. Discussion

With the quantitative results established, and the impact of sponsorship on a student-level estimated, it remains to place the impact of the programme back into the context of other educational programmes in the Global South. This section also explores the mechanisms through which sponsorship may be an effective programme, largely drawing on the semi-structured interviews with the different stakeholders of the programme. Then the wider applicability of these findings are discussed. Finally, some thoughts on the usefulness of student level data in measuring educational impact are offered.

7.1. Relative Effectiveness of the Link Ethiopia Programme

As discussed earlier the effectiveness of an educational programme, should not be seen in isolation, but in comparison to other programmes. This can be effected by comparing the improvement (or otherwise) in the outcome variables of the Link Ethiopia programme with other studies and programmes. Such an approach, whilst comparing outputs, often says little about effectiveness in relation to inputs, and thus it is also crucial to compare the impact of the programme in terms of cost-effectiveness. The majority of evidence suggested that there were increased benefits to longer periods of sponsorship. These models (Model 3 and Model 4) are useful as they allow the prediction of the expected outcomes for students, where the model holds, based on the cumulative months of sponsorship that student has received. The average length of Link Ethiopia sponsorship at the end of the 2014/15 academic year was 46.2 months.

For the variables of enrolment and promotion the relationship with cumulative months of sponsorship is non-linear, and on average sponsored students perform below the parallel trend during the first years of sponsorship. Then at a turning point the average effect begins to increase back towards and then above a net average effect of 0. This may be due to self-selection within the data where for a certain subset of students, in spite of sponsorship, they perform significantly below the parallel trend, dragging the average effect down. These students then dropout, and the average effect of sponsorship begins to increase, reflecting the positive impact of

sponsorship on a different subset of students. It should be noted however, that when only looking at students who are currently enrolled, the relationship between sponsorship and promotion remained non-linear, suggesting that this relationship is not be entirely contributable to self-selection.

Two other possible explanations for this non-linear relationship are time-lags to the benefits of sponsorship, or changes in the programme design. Time-lags may be due to the vulnerable situations of the sponsored students and their families, such that it takes time for sponsorship to reliably make attending school financially viable. Alternatively, the increase in expenditure on individual support from the start of the 2012/13 may explain this non-linear relationship, especially as the programme has not been established long enough for the start date of sponsorship and the length of sponsorship to not be correlated. Educational outcomes may have performed below the parallel trend when only 20% of charitable expenditure was allocated for individual support, but since this has increased to 55% sponsored students may have performed above the parallel trend. It is entirely possibly that a combination of self-selection, time-lags, and increased individual support are all at play simultaneously.

Finally, it may also be possible that the impact on the sponsored students is not fully captured by the difference-in-difference analysis. When looking at the descriptive statistics of the two groups, the non-sponsored students have considerably lower enrolment and promotion rates than the sponsored students. This may imply that, for those sponsored students who do remain in education, the data prior to intervention is suggesting that those students would have remained in and progressed through the education system even without sponsorship. Prior enrolment and promotion. For example, a student may be able to comfortably attend education, but then experience a shock such as a family death and a subsequent fall in income. The student may then become eligible for sponsorship, but, in such a case the full impact of the sponsorship on enrolment would not be represented in the estimated programme effect.

Based on the average length of Link Ethiopia sponsorship of 46.2 months the average impact for each student by the end of 2014/15 can be calculated. For enrolment, if the student-fixed effects are used, then the enrolment rate for the average student has decreased by 5.8 percentage points, and if the school-grade fixed effects model is used then this decreases by 3.2 percentage points.

Irrespective of the underlying cause of the non-linear relationship, the model predicts cumulative exponential benefits to sponsorship. So if the average length of sponsorship increases, the programme can have a net beneficial impact on enrolment. Given that the scheme is still relatively young, with many students only having been sponsored for a short length of time so far, providing Link Ethiopia can keep students in the scheme and in school, then that average length of sponsorship will increase.

Analysing the PROGRESA programme estimates that receiving the CCTs for 5.5 years increases the years of schooling by (0.8-1.0 years) (Behrman et al. 2011). If Link Ethiopia sponsorship was received for 5.5 years (66 months) then under the student fixed effects model the enrolment rate would have reduced by 1.76 percentage points, but using school-grade fixed effects the enrolment rate increases by 0.98 percentage points (.05 years of schooling). The Wydick et al. (2013) analysis of Compassion has students being sponsored for an average of 111.6 months. Estimating the effects of this length of sponsorship on enrolment rates increases enrolment rates by 22.4 percentage points (2.08 years of schooling) using student fixed effects. Of course, enrolment rates have a ceiling of 100%, and thus the effects might not be quite so large, however it does serve to highlight the point that, if the scheme continues, there is potential for large increases in enrolment due to the scheme.

The same analysis can be carried out when examining the effect of sponsorship on the promotion rates, which also follow the cumulative non-linear model. Again using the average length of sponsorship of 46.2 months, the estimated effect of the programme is a reduction in the promotion rate by 1.10 percentage points for student
fixed effects, and 2.22 percentage points for school-grade fixed effects. When estimating the impact for 66 months of sponsorship, promotion rates are increased by 9.87 percentage points, or 10.28 percentage points for student and school-grade fixed effects respectively.

The relationship between the sponsorship and grade ratios was found to be more simple, and calculations can be based on the result from the linear relationship between Grade Ratio 2, which is a composite measure of enrolment and promotion, and cumulative months of sponsorship. Using this and the average length of sponsorship, the average student can be seen to have a grade ratio of .087 points (out of 1), higher due to sponsorship. If years of schooling are considered to represent the years progressed through the system, rather than mere attendance, then after 42.6 months, sponsored students have 0.31 additional years of schooling.

Based on the sponsorship budget, 42.6 months of sponsorship provides £292.88 of individual educational support for a student. Using this estimate and the additional years of schooling based on Grade Ratio 2 the scheme can currently be seen to cost £944.75, approximately \$1239 at current exchange rates for an additional year of schooling. As the Link Ethiopia model only changed from a 20/80 individual support/school fund split to 55/45 split for the start of the 2012/13 academic year, the actual expenditure is much lower than this. According to the Link Ethiopia financial accounts, including overheads contributable to individual student support, a total of £16,299 has been spent supporting individual students. This computes to a mean expenditure per student of £168.03. With the average length of sponsorship still being 42.6 months, the cost per additional year of schooling is £542.05 (\$710.79). If the overhead costs of the programme are removed, then this estimate falls further to (£391.53).

In terms of cost-effectiveness this puts the programme in a similar region as the PROGRESA programme (\$614 per additional year of schooling), but much less costeffective than other Sub-Saharan educational programmes such as uniform distribution (\$91) or deworming (\$3.50). These studies based their calculations on the attendance at school, rather than progression through the education system, and

considering this as a direct comparison is therefore, not completely accurate. The lack of reliable attendance data from Ethiopian schools limits the ability to compare this study to others that measured the impact on attendance. It is also possible that since the uniform study, inflation may have increased costs in the last 10 years, and improvements to education systems may have also increased the marginal costs of an additional year of schooling. The significantly higher costs of the sponsorship model, compared to mere uniform purchase, highlights how much the uniform is an important aspect of increasing access to education. Finally, with greater improvements in educational outcomes predicted, merely by an increase in the average length of sponsorship, the cost-effectiveness of the programme seems set to increase with time.

7.2. Mechanisms of Sponsorship

The semi-structured interviews shed light on three main ways in which sponsorship can impact on educational outcomes. They demonstrate how school resources and CCTs both reduce the private costs to, and therefore incentivise, education; how the provision of those resources is more important than just as mere incentives; and finally they highlight the role that emotional support and the raising of aspirations can play in improving educational outcomes.

Students consistently reported that the provision of uniform and exercise books positively impacted their ability to attend school, and expressed the view that the cost of these resources is a barrier to education. When one student was asked what would happen if they didn't receive exercise books and their uniform they responded, *'If I don't have exercise books and uniform, then sometimes the food money will go for exercise books so maybe some days we will not have food'*. Alternatively, another student stated that their guardian, their aunt, was delighted about her receiving sponsorship saying, *'it is enough for your school so I can help you with the rest of your things'*. Both students are expressing here how the private costs to education are a real factor in family decision-making of whether a child attends school, and how sponsorship reduces this cost and incentivises education. One Kindu student also cited the CCT as a major reason for continuing to attend school, stating *'I love to go to school, because I want to help my parents, because they get support from*

the organisation', and another student stated that *'the money'* was the most important aspect of Kindu support to their mother.

One student expressly stated that the lack of uniform was the main reason that they didn't attend school until the age of 14, due to it being a mandatory requirement of the school, and the families' poverty being the reason that she didn't have a uniform. Likewise, the school link coordinator suggested that exercise book provision was important because *'they cannot buy exercise books they cannot learn, it is important*', i.e. without the ability to write in class and take notes and do exercises then a student's ability to learn was particularly limited. Thus, as well as reducing the private costs of education, the provision of exercise books promotes learning and contributes to the student's ability to progress through the education system.

The final mechanism that sponsorship may be effective through is the raising of aspirations, and the positive experience of being supported through education (Glewwe et al. 2013). Sponsored students expressed very high levels of educational aspiration, with expressions such as 'I have to study before the examination, so it's my passion to go to university so I will succeed', and 'If I get a good score then I will go to university and then I will get a job', which also expressed a need for determination and hard work to study well to attend university. Of course, these may be the answers that would be expected from students, irrespective of whether they were sponsored or not, but such responses correspond with Glewwe et al.'s (2013) guantitative analysis, that found higher aspiration levels of sponsored students compared to non-sponsored students. Given that Link students are means-tested and from low socio-economic backgrounds, these responses were very encouraging. As well as expressing such aspirations, students also expressed how being sponsored contributed to this. When asked about how it felt to receive sponsorship one student stated 'I was so happy', and another emphasised the importance of emotional support in education, 'I must be happy. If you have a person who helps you, you try harder and harder.'

This anecdotal evidence provides qualitative evidence to back up the various claims about how education is perceived on a cost-benefit basis, how poverty is a barrier to

education, and how changing the costs of education affects the decision to attend school. The interviews also explain how exercise books and uniforms are particularly important inputs into the educational system, as well as showing how sponsorship can support students emotionally and raise educational aspirations, and how these mechanisms operate to improve educational outcomes. The consultancy report provides further insights from the qualitative interviews.

7.3. Generalisability of Findings

As already mentioned, the existing literature on the impact of educational programmes in the Global South, let alone child sponsorship programmes is extremely limited. The use of individual level data to estimate the impact of supporting individuals in their education is even more so. Thus the finding of improvements in individual educational outcomes with increased sponsorship duration is difficult to corroborate with other literature. Given that the mechanisms of sponsorship identified in the semi-structured interviews correspond with other studies of educational programmes in the Global South; uniform (Kremer et al. 2003), CCTs (Schultz 2004; Behrman et al. 2005; Behrman et al. 2011) and raised aspirations (Glewwe et al. 2013), this result of improved outcomes with time could be generally applicable across sponsorship schemes that provide similar support. Given the wide variety of sponsorship schemes however, including public goods models, this should not be considered applicable to all models of sponsorship.

The non-linear relationship between the length of sponsorship and educational outcomes is less likely to be widely applicable. Given the wide range of possible factors driving this relationship, including reasons that are specific to the Link Ethiopia programme such as the change in sponsorship model, there is little evidence to suggest that such a relationship is inherent to child sponsorship models. This example does though, serve to illustrate how the selection of students for a sponsorship programme can greatly affect the impact of the programme. If the programme is either insufficient for some students, or not required by others then the programme is not effective. Alternatively, the programme may be effective, but if students become eligible for sponsorship due to a shock, then the effect of the sponsorship programme will not be captured by difference-in-difference analysis.

Finally, consideration should be given to how geographically generalisable these findings are. Firstly this study focussed on urban schools and students, and in Ethiopia rural students have lesser access to education (Ministry of Education 2015a). Likewise, within Ethiopia there are large regional differences in educational outcomes, and Amhara region, the site of this study, consistently performs above the national average (Ministry of Education 2015b). Thus care should be taken in considering whether such findings are representative of either Ethiopia or Sub-Saharan Africa. However, given that the provision of uniform has been found to be effective in Kenya (Evans et al. 2009) and the raising of aspirations through sponsorship was found in Kenya and Indonesia (Glewwe et al. 2013), the positive impact on educational outcomes of sponsorship and the mechanisms underlying sponsorship identified will be applicable well beyond the Amhara region.

7.4. Usefulness of Student-Level Data

The availability of student-level data, often stretching back many years, can be seen as useful source of data for evaluating educational programmes in Ethiopia. Studies such as Wydick et al. (2013) often rely on estimation strategies, based on grouplevel data on educational outcomes, and the probability of an individual's exposure to a programme, to measure programme effects. By using actual student-level data there is greater confidence in the accuracy of the outcomes of students for programmes that are focussed on the individual, and this approach also allowed for the inclusion of student-fixed effects, compared to household fixed-effects in the analysis of the Compassion programme.

Given that the data is available historically there is potential usefulness for other individual-level programmes and studies using difference-in-difference in analysis. As Glewwe et al. (2004) argued though, prospective RCTs can produce more accurate results, and this data would be very suitable for such an approach, whether looking at individual or school-level programmes. Carrying out prospective analysis would also allow for more certainty in the eventual of outcomes of students, as to whether they have dropped out or changed school for example, which would provide more reliable data, and more accurate results. The only other concern with the data as it is currently managed is that attendance data does not seem to be collected

consistently across schools. Finally, noting that often the outcomes for students were unknown, and that often students were enrolled with little evidence of attending school, it suggests that collecting data at an aggregate level, whether that is at a district, regional or national level is an extremely challenging process.

8. Consultancy Report

This section is primarily targeted towards Link Ethiopia, to provide useful insight into the effectiveness of the sponsorship programme, and the implications of this for the programme and organisation as a whole. This section presents a summary of the findings that are of interest to Link Ethiopia followed by recommendations for improvements to the programme, based on the findings from both the quantitative and qualitative data. The report finishes with suggestions on how the monitoring and evaluation of the programme can be improved, with suggestions for future research on the programme, and any implications for the monitoring and evaluation policy of the organisation.

8.1. Summary of Main findings from the Quantitative Analysis

The majority of evidence suggests that the benefits from the Link Ethiopia sponsorship programme increase as the duration of sponsorship increases, and that in its current form, the sponsorship programme is only beneficial to a subset of students. At the end of the 2014/15 academic year, the average length of sponsorship was 42.6 months and therefore the net average effect per student was close to 0 for both enrolment and promotion rates. In the same time period however, the scheme was responsible for improvements in Grade Ratio 2, a composite indicator of enrolment and promotion. This indicated an average increase of 0.31 additional years of schooling for sponsored students over this period, and should be considered a success by the organisation.

Based on the actual expenditure of the organisation, the estimated cost of an additional year of schooling is £542 (including overheads) or £391 (not including overheads). This suggests that the programme is less effective than others that take place in similar geographic and socio-economic contexts, such as those that focus on uniform provision (Kremer et al. 2003), or deworming programmes (Miguel & Kremer 2004). The Link Ethiopia sponsorship programme is more similar in cost-effectiveness to the PROGRESA programme of CCTs in Mexico, a middle-income country.

There are many possible reasons for this higher than expected cost of an additional year of schooling. For one, the method of measurement has been different for this study, due to unreliable data on attendance. Secondly since those studies, the marginal costs of additional years of schooling may have increased. Thirdly and perhaps most importantly, by increasing the average length of sponsorship Link Ethiopia can expect to see rapid improvements in average effects per student across educational outcomes for its sponsored students.

This assessment is based on the indication that the relationship between sponsorship duration and both enrolment and promotion rates is non-linear. This means that the average effects per student for these indicators decline in the first years of sponsorship before improving. Whilst the average length of sponsorship is currently just short of the required length to return the average effects on these indicators back to 0, a small increase in the average length of sponsorship will achieve this, and thereafter average effects will continue to increase. An increase in the average length of sponsorship, whilst not inevitable if students continue to dropout, is certainly plausible, the average length of Compassion sponsorship was 9.3 years (Wydick et al. 2013), longer than the Link Sponsorship scheme has even been in existence.

The reasons for this non-linear effect warrants serious consideration by the organisation. One possible cause is that within the students who are selected for sponsorship there are some students for whom the sponsorship programme is not sufficient to maintain their education, and therefore those students' outcomes continue to decline and they then dropout within the first few years of sponsorship. Thereafter, when those students have dropped out, the average effect per remaining student for both enrolment and promotion is positive, and thus for these students the sponsorship programme can be seen to having a positive effect.

Given that the descriptive statistics show higher promotion and enrolment rates for sponsored students compared to non-sponsored students, but the difference-indifference analysis found limited programme effects on enrolment and promotion, this would suggest that, prior to intervention, that some of the students selected for

sponsorship may have had good outcomes for enrolment and promotion. The organisation may wish to reflect on whether the students selected would have continued with these good educational outcomes in the absence of sponsorship.

It may be that when the selection process was carried out by schools, there was a subconscious bias towards students who were performing well in school, within that subset of eligible students. Link Ethiopia should consider whether this may be the case, and if it is, whether those students were still at risk of dropout in spite of strong educational outcomes prior to sponsorship. One such example where this may be the case is if a student should suddenly become eligible for sponsorship due to a shock such a parental death. Their educational outcomes may have been good prior to sponsorship, and therefore the impact of sponsorship will have not been captured in the difference-in-difference analysis.

Alternatively, given that at this early stage the months of sponsorship is linked to the amount of time the programme has been in existence, Link Ethiopia should also consider whether this non-linear relationship reflects the effectiveness in the programme over time; does the organisation feel that the programme may well have been ineffective at the start, and has now improved and become more effective? In particular, the change in model from 20% to 45% of charitable spending being introduced in 2012/13 could be a significant factor.

Link Ethiopia should also consider whether sponsorship simply takes time to be effective, as this would also explain the non-linear effect. It may be that there are significant time lags between the provision of resources and impact on educational outcomes. For such a lag, however, to be more than a year or two, as the results suggest it would have to be to entirely explain the non-linear relationship, seems unlikely.

Importantly, whatever the cause of the non-linear relationship, the quantitative analysis predicts that the biggest improvements in educational outcomes across all indicators occur when a student is sponsored for a long period of time. It may take 6 years for the average student to see improvements in all educational outcomes, with outcomes declining in the first years before improving. This means that Link Ethiopia

must view individual sponsorship should as a long term commitment to students, rather than as an intervention that has immediate effects.

The other key takeaway from the research is that there is little evidence that CCTs through the Kindu model are a more effective mechanism for improving educational outcomes. It should be borne in mind that there is some evidence that the CCT improves attendance in relation to non-sponsored students, more than under the Link Ethiopia model, but the data available meant that carrying out analysis on the effect of the CCT in preventing dropout was not possible. There is also anecdotal evidence from the interviews that CCTs are an effective incentive to education.

There is no evidence though that the CCT improves progress or attainment beyond what the Link programme offers. On the contrary, the evidence, whilst inconclusive, suggests that the Link Ethiopia programme is more effective in improving progress and attainment than the Kindu programme. Given the lower cost of the Link scheme, the programme can also be considered to be more efficient in improving educational outcomes. Of major consideration however, is that this analysis does not account for the benefits of the Kindu model to the family, particularly in relation to the CCT and medical support.

Finally, for Link Ethiopia, it is relevant to consider the effectiveness of the individual support provided to students, in relation to the effectiveness of its other programmes. So far to date the sponsorship programme in Gondar has cost £11,733 in individual students support and £34,925 has been raised for the school improvement fund (not including overheads in either figure). In total, based on the Grade Ratio 2 calculations, this individual support has provided an additional 30 years of schooling between the 97 students supported. It is recommended that the organisation carries out further analysis to compare this return to the impact of its other programmes.

If its other programmes are considered more effective, then the emphasis of sponsorship should be on raising funds for the educational environment improvement fund. Of course, if it is felt that the non-linear relationship between sponsorship duration and educational outcomes is due to the increase in individual support, then this would not be an efficient use of resources, unless the programme

was to move entirely to a school-level public goods model. If Link Ethiopia considers the returns to the sponsorship programme to be more efficient than some of its other programmes, and wants to focus more on individual support, more resources to students may not necessarily lead to improved educational outcomes. It may be more efficient to use the money raised for the educational environment fund to sponsor one additional student. With economies of scale to be gained through reduced overheads of sponsoring more students, two students could probably be sponsored through the £15 provided monthly by sponsors.

8.2. Programme Recommendations

Some of the recommendations to improve the programme stem from the quantitative analysis, which have highlighted some of the weaknesses in the programme. The vast majority of evidence however, for the recommendations, is drawn from both the semi-structured interviews and observations from my time in the Link Ethiopia office.

The main takeaway from the quantitative analysis was that for the average student in the programme most of the educational outcomes tend to get worse before getting better. Accepting that students tend to dropout in the first few years of the sponsorship scheme suggests that if this issue can be addressed by the programme it will bring about dramatic improvements in its effectiveness, particularly given the improved outcomes that are evident after this stage. The two areas that warrant consideration are the selection process, and ways to improve school retention in the first years of sponsorship.

8.2.1. Improving the Selection Process

The quantitative analysis has yielded two main possibilities about the sponsored students. The first is that it is possible that within the group there may be students who are being sponsored for whom the sponsorship is not sufficient to prevent their dropout. Link Ethiopia currently struggles to contact students who have dropped out under the scheme, as evidenced by the difficulty of getting interviews with such students. When students drop out, more needs to be done understand why they are not staying in their education, to build up a bank of evidence about how to improve the scheme.

Secondly, there is the suggestion that, within the sponsored students, there are certain students receiving sponsorship who would have stayed in school anyway. This may be attributable to the selection process initially being in cooperation with schools, rather than the Women's and Children Association (WCA). Link Coordinators and School Directors may have favoured students within the group of eligible students who had stronger educational outcomes prior to intervention. It is also possible that, for some of the students, they may have become eligible for sponsorship due to a 'shock', such as the loss of a parent, which dramatically alters their circumstances. If this was the case, then such students would have required sponsorship, but this is not reflected in the difference-in-difference analysis.

Now that the selection process is carried out with the WCA rather than schools, this may help to reduce selection bias in the system. Given the findings of this research however, Link Ethiopia should not expect that, simply because a student is eligible for sponsorship they are an appropriate candidate for the sponsorship programme. It is recommended that Link Ethiopia arranges a time to meet with the WCA to better understand the selection process, with these concerns in mind, to be more certain that for the students selected, the sponsorship is both required and sufficient. If Link Ethiopia develops strict criteria for its selection process, on the basis that it will boost the effectiveness and efficiency of the programme, they may want to consider if this will conflict with principles of equity, fairness and equality of opportunity. Limiting the sponsorship programme based on the expected impact of sponsorship of certain indicators may remove the opportunity for some children to access sponsorship who could potentially benefit from the support.

8.2.2. Understanding the Situations of Sponsored Students

The interviews with students helped to shed some light on the individual situations of sponsored students, and whether they were in a situation where sponsorship would be able to help them. Two students in particular mentioned that the inability to afford a uniform was a major issue, with their first uniform paid for by either a teacher or a distant relative. Many Link Ethiopia students also mentioned that they often also had part-time jobs outside of school. Older boys often reported working as mechanics in their spare time, and girls reported helping their parents/guardians carrying injera as

part of the family business, suggesting that many sponsored students also had to bring in household income. Whilst this is not uncommon in Ethiopia, it certainly suggests that students selected for sponsorship were not from comfortable or 'middle-class' backgrounds, nor that sponsorship elevated students into such a category.

Other reported professions of parents or siblings of sponsored students included shopkeepers, guards, minibus ticket collectors, waitresses and bajaj drivers. Such jobs, would all be classified as low-skilled and low-waged. For most students however, there was evidence of some sort of income to the family. Given the high levels of unemployment in Gondar, this may suggest that the students were not from the most impoverished families in the town. The organisation may wish to consider if they want to reach such families, which probably corresponds to students who are not in school to begin with, or whether given the concern that sponsorship may not be sufficient for some families for the student to remain in education, that families with low-wage income are the appropriate target group.

There was also a suggestion from those students who were eligible for sponsorship due to being orphaned that this was understood by their peers, *'most of the students depend on their family and get support from their family. So I am orphan, so* [Kindu] *just like a family. They not insult me because, when I receive from The Kindu Trust, the issue of family. It's equal'.* This suggests that the priority given to orphans is justified, in that to attend education students have to be reliant on their parents. Students who had lost one or two parents however, often reported living with older siblings, with those siblings providing an income for the family. Again whilst these sources of incomes were generally from low wage jobs, income from siblings and not just parents/guardians should be included in any means-assessment. Likewise, given that employment situations can be fluid, such as an older sibling graduating from university, means assessment should be repeated at regular intervals to ensure that the student is still eligible for sponsorship.

Interviews also explored with students what would have happened to them in the absence of sponsorship. As mentioned previously, sponsored students had very high

educational aspirations, and coupled to this was a determination to succeed in education, '*I have to study before the examination, so it's my passion to go to university so I will succeed*.' Students showed a clear determination to stay in education, and when one student was asked what their family would say about their education in the absence of Link Ethiopia support stated '*They would say attend your education*.'

Despite this expression of aspiration and determination many students voiced that in the absence of sponsorship they would ultimately dropout from their education, stating it would be due to an inability to purchase resources, '*No one will facilitate my exercise books*'. It may be considered that this is the expected response from sponsored students, however other, more nuanced opinions were voiced. The Link Coordinator expressed that within the sponsored students there is variation in their situations, and that some would dropout, whilst others wouldn't. Another student voiced that school resources would be forgone to purchase food, even though their mother '*would not be happy about it*'. A further student thought that whilst they would still attend school, they would be repeatedly detained (repeat grades), as the lack of school resources would limit their ability to learn. Finally, another student suggested that if sponsorship did not exist, whilst they may not have dropped out immediately, it would have happened eventually, stating it would have happened about 4 years after the sponsorship started, as this was when their older brother went to prison, and this affected the family's income.

So whilst it is not clear cut that for all students that are part of the scheme, that sponsorship is the difference between attending school and dropout, the interviews suggested real trade-offs between education and other goods, and that sponsorship had had a positive impact on the educational outcomes of those students.

8.2.3. Improving Student Support in the First Years of Sponsorship

If Link Ethiopia is ultimately satisfied with the selection process, then it must consider ways to improve the support provided to students in the first years of sponsorship. The analysis suggests that the first 2-3 years is when the sponsored students are most at risk of dropping out. After this point the risk of dropout and repetition of

grades is reduced. The organisation should think about the most appropriate support that can be provided for students in these years, whether that is additional incentives that are available for the first three years of sponsorship, or increased home visits to offer emotional support and raise educational aspirations. Increased home visits also improve the monitoring in this period, and monitoring efforts should be focussed on this phase of sponsorship. More regular monitoring of the students' needs through this period, and possible use of the contingency fund during this period could be a good use of resources. Finally, given that the benefits of sponsorship seem to increase with time, it would also be prudent to sponsor students in the earlier stages of their education. This allows for the longest possible period of sponsorship and should maximise the benefits from the programme.

8.2.4. Relative Importance of School Resources Provided

Link Ethiopia may also want to consider the relative importance of the types of individual support that students receive. Link Ethiopia students were asked to rank in order of priority the different resources they receive including the sponsorship clubs. Nearly all of the students interviewed stated that uniforms and exercise books, and school resources in general were the most important items they received. Of the remainder, typically, the school bag was ranked next, followed by shoes, and lastly the sponsorship clubs. Only one student stated that the bag was more important than the uniform. The reasons for the importance of the exercise books and uniforms have already been outlined previously, and in terms of efficiency and cost-effectiveness Link Ethiopia may need to consider whether the provision of the school bag and shoes is necessary, especially when exercise books and uniforms seem so highly valued.

8.2.5. Improving Sponsorship Clubs

The sponsorship clubs seemed to largely be ranked last due to a lack of awareness from students about the sponsorship clubs. One sponsored student stated that they were not an active participant at the club, and did not know what activities happened at the club. The Link Coordinator also stated the sponsorship club was merged with another club, perhaps suggesting that it wasn't getting the attention that it deserved. They also felt that the sponsorship clubs should be used to improve the spoken

English of students. One student who said that there was not a sponsorship club at their school stated that if there was and it focussed on teaching English that would be very helpful. The student even went so far to suggest that if there was a straight choice between receiving shoes and getting additional tuition then they would choose the tuition. This would suggest that whilst sponsorship clubs are not currently running efficiently, if they were to do so, they would be greatly welcomed by the students. The clubs should focus on additional tuition, and it should be a matter of priority of Link Ethiopia to establish how the clubs are being run in the various schools.

8.2.6. Importance of Emotional Support

The interviews also highlighted the important role that emotional support can play in a student's education. One student said that peer pressure could be very distracting for teenagers, and therefore the organisation should constantly interact with its students to help build their confidence and persuade them to stay in education, rather than seek work. The Link Coordinator, when mentioning another NGO in the region put a strong emphasis on the importance of home visits. These interviews suggested that it was important that a student feels that someone is directly interested in their education. Whilst home visits are time-consuming, and therefore costly, there is a strong suggestion that this emotional support can improve educational outcomes, and so ways to interact more with the students should be sought. This could be achieved by enhancing the sponsorship clubs, giving the Link Coordinator a greater role in the provision of welfare for sponsored students, but also through Link Ethiopia staff attending sponsorship clubs and interacting with students.

8.2.7. Possible Additions to the Sponsorship Scheme

If Link Ethiopia was considering adding or substituting something into the programme, food support would be an option. This paper has already mentioned studies that show that provision of food can be a strong incentive for education, and the provision of food was voluntarily raised by multiple stakeholders. Students often referred to lack of food at home being a priority concern, that may affect their education, and as the Link Coordinator put it, *'If they have school materials they can learn, but if they don't have enough to eat they cannot come to school'*, suggesting

that not only would food provision incentivise education, but that it was in fact a precursor to education, and that hungry students would simply not attend school.

The seemingly cost-effectiveness of the deworming programme at \$3.50 might make this an attractive option for Link Ethiopia. Given the potential benefits this possibility should certainly be explored, however Ethiopia has recently launched a national deworming programme (Weldon 2015). Link Ethiopia should seek to establish if its sponsored students are at risk of worms, and if they are whether this need would be met by the government. It may be that there is no role for Link Ethiopia if such a programme sufficiently met the needs of sponsored students, however it may be that Link Ethiopia can help its students access a deworming programme, or consider if it should expend its own resources on incorporating deworming into its sponsorship programme.

8.2.8. Other Recommendations from Interviewees

Interviewees were also asked to suggest any possible improvements to programme. As mentioned, food support, and a lack of food was raised by students and the link coordinator. The Link coordinator also suggested income support, stating that a simple lack of income was a cause of dropout, with students taking on jobs such as shoe cleaning instead. Of course, having a part-time job, and staying in education are not mutually exclusive, unless that job becomes full-time, or takes up so much time that there is no time for studying outside of school hours. Finally, students also highlighted that in the Kindu programme, inflation mean that the CCTs were reducing in value, and also suggested that whilst the support was sufficient for those in school, it was not sufficient for those attending university. Perhaps the most interesting recommendation came from one student who suggested, that if a student finishes education prior to attending university, then the option for a loan to start a small business should be considered as the final support offered by the organisation. This could either be in place of, or come after attending vocational college.

8.2.9. Other Issues Explored in Interviews

The interviews also raised other unrelated issues, that may be of interest to the organisation. Students were asked to consider whether they discussed sponsorship

with their friends, if people knew they were sponsored, and how they felt about it. Some students had no issue with being known as sponsored students, often citing how the organisation simply supported their education in a way their family could not. There was little evidence of any student feeling that friends of theirs begrudged their sponsorship, simply describing it as *'lucky'*. This acceptance of sponsorship by peers was also attributed to friends not making them feeling *'anything'* about being sponsored.

One student specifically voiced that they would not want someone who was not their friend knowing that they were sponsored as, *'If someone says from outside I will feel sad, but the organisation does good things for me, so I will not feel sad, but for a moment I will feel embarrassed. If someone knows that I get help, or if his parents are rich then they will know this. People will think that you are poor'.* This illustrates that there is a stigma associated with poverty and therefore sponsored students could be stigmatised. This suggests that the organisation should always be mindful of ways in which sponsorship can remain as anonymous as possible. For example, sponsorship clubs could not be named as such at school, and resources could be distributed at homes rather than in school. Many students did however report not discussing sponsorship with their friends, and thus it doesn't seem that it is an issue that students feel a need to disclose or are questioned about.

8.2.10. Merging the Link Ethiopia and Kindu Sponsorship Programmes

Given that there are essentially two different sponsorship schemes being run by two different organisations, but these two organisations are due to merge there seem to be many ways in which the schemes can be combined to become more efficient. The evidence would suggest that the Link model is more efficient, in that it is producing better educational outcomes with a lower expenditure. This comes with the caveat that analysis on how much the CCTs improve access to education was not possible. More research should be done on exactly how effective the CCTs are in increasing enrolment, however the research did clearly suggest that the CCT does not have an additional positive effect on academic progress or attainment, when provided on top of school resources. Given that there is a concern with the Link Ethiopia selection process, in that for some students the provision of school

resources is not sufficient to prevent dropout, a unified selection process between the Kindu and Link programmes could have great benefits.

Assuming that CCTs do have positive outcomes on enrolment, the Kindu programme could be targeted at students from the poorest families, where the CCT can make the difference between attending school and dropout. It may also be that the Kindu programme can focus on students who are currently not part of the education system, whereas the Link Ethiopia programme can focus on students currently in school, who are identified as at risk of dropout. Given that the Link Ethiopia programme has so far selected from students in school, and some of these students have dropped out, it may be that in fact students currently in school are the most appropriate for both programmes, but within this group, students from the poorest families are more appropriate for the Kindu programme, and other students are more appropriate for the Link programme. For this to be efficient, both programmes would have to carry out the same method of means-testing, and to have an agreed set of criteria which determines the most appropriate scheme for the student. This coordination between the two schemes would also mean that, should through regular mean-testing, it be indicated that the financial situation of a student on the programme had improved, they could be moved from the Kindu programme onto the Link Ethiopia programme.

By merging the two programmes there are also efficiency gains to be made. Some gains are straightforward to implement. For example, aligning the resourcing cycle of both organisations should lead to reduced costs when purchasing school resources and uniforms due to purchasing larger quantities of both simultaneously. There should also be gains in the human resources expended in the distribution of resources process. Currently the two organisations are separately spending time visiting homes and schools to distribute resources, which is a time consuming process. The data showed that students from both programmes attend many similar schools, and therefore live in similar areas, meaning that this process can become more efficient. Going forward, focussing on the same schools and areas for both programmes can also improve efficiency, creating more time for home visits, and should reduce the costs when purchasing uniforms for each school.

Given the emphasis in the interviews placed on the importance of providing emotional support, one of the main strengths of the Kindu programme is the emphasis it places on home visits. The Kindu programme also employs trained social workers, and this expertise could enhance the Link sponsorship programme through increased home visits. By Kindu social workers combining more extensive home visits with resource distribution this can also reduce the resource distribution tasks of the Link Ethiopia sponsorship coordinator. The Link Ethiopia sponsorship coordinator could then focus on more on programme management, monitoring and evaluation, and coordinating the two schemes with the Kindu sponsorship manager.

The strength of the Link Ethiopia programme is that it has much stronger relationships with the schools its students attend than The Kindu Trust does. This relationship with schools can be used more efficiently than it currently is. It may be more efficient for Link Coordinators to take on greater responsibilities in the sponsorship programme. As well as helping with resource distribution, they can also assist with gathering information and photos for sponsorship updates and sponsor feedback. Given that sponsorship clubs seem to currently be ineffective, Link Coordinators could also take on this responsibility, providing extra tuition and being appropriately compensated through the sponsorship clubs. The clubs could be open to Kindu students as well, and the Link Coordinators could also use the sponsorship clubs simultaneously to provide that additional emotional support and motivation. The clubs would form an integral part of the monitoring of the programme, highlighting problems faced by students and flagging any students struggling in their education and referring the case to Link Ethiopia. There are also gains to be made in a unified approach to monitoring and evaluation process between the two organisations, which is outlined in the following section.

8.3. Improving Monitoring and Evaluation of the Programme

It is hoped that this study has laid the foundations for improving the quantitative evaluation process, through exploring the possibilities of data availability, and highlighting the possible form that analysis of the programme can take. The major outcomes identified: enrolment, promotion, days absent, average result (from internal exams), and the calculation of both grade ratios used in this study, can form

the basis for the organisation's approach to monitoring and evaluating the programme. As mentioned, the only unreliable metric of these is days absent, due to the inconsistent way in which this is calculated across schools. By recording such data on an annual basis, in future, the data will be more reliable than when collected retrospectively. Likewise, when done annually this collection and entering of data can be incorporated into the end of year sponsorship update as all the data required should be available as part of the students 'report card' from the school.

The collection of such data can significantly add to the monitoring process of the scheme. It makes it straightforward to identify when a student is struggling, repeating grades, performing badly in end of year of exams, or often absent from school. This will enable such cases to be investigated, and solutions to be found, improving the retention of students in education. In terms of effective evaluation, continued data collection on random students is required to maintain the control group. This would require cooperation with the schools involved, however this initial data set could provide an initial list of students to continue to track on an annual basis, and could be built on by each year simply adding five randomly selected students from Grade 1 at each school to maintain the control data set. By collecting this data annually, outcomes that in the historic data that were unknown, such as dropout or a change of school would be known, improving the reliability of the data and the accuracy of the analysis. Not only would this allow the repetition of this analysis on an annual basis with the updated data, but this data may also be useful beyond the sponsorship, as discussed in the next section.

As has already been suggested, the unification of the selection process between the Kindu and Link programmes, means-testing and collection of baseline data can improve the effectiveness of the organisation. Attached to this process should be the collection of the previous academic history of the sponsored student. This information will help the organisation better understand the student being sponsored, the type of support they require, and be important to any future analysis of the programme.

8.3.1. Further Research

This section concludes with some suggestions about further research that could be carried out to improve the monitoring and evaluation of the programme. Firstly, as this study was only based on Gondar students, the same methodology could be applied to students supported through the Debre Zeit office. Different results may be found due to the regional variations that mean that Oromiya performs significantly worse than Amhara across all educational indicators (Ministry of Education 2015b).

More research could be undertaken to understand the selection process. To investigate this quantitatively, information on the socio-economic data of students, from both Kindu and Link could be linked to the educational data, to identify which students benefit the most from sponsorship. Even with the existing data set, further analysis could be run to see if the impact of sponsorship is gendered, or benefits students more who first receive sponsorship in a given grade or at a certain age.

Using the existing data, the educational outcomes of sponsored students prior to sponsorship could be compared with the non-sponsored students. This would help Link Ethiopia understand if the selection process currently favours students with either better or worse educational outcomes. Similarly, this analysis could also categorise the sponsored students based on their educational outcomes prior to sponsorship, to see if there are certain educational indicators that suggest that a student would benefit more from sponsorship than others. This analysis could highlight whether well-performing or poorly-performing students benefit more from the programme. By including academic history as part of the selection process, Link Ethiopia may be able to increase the impact of the programme.

Finally, tracking of all students who are ever part of the scheme, into the future is important, particularly recording the level of education that the students finally complete. This would allow Link Ethiopia to compare the impact of its programme against the same outcomes explored in the Compassion International impact evaluation (Wydick et al. 2013) at some point in the future when enough students have 'graduated' from the scheme to make the analysis viable. Tracking students for a length of time can also allow for improved qualitative analysis of the programme,

as students who have been through that journey can provide great insight into the challenges they faced, and reflect on the ways in which sponsorship did, and did not help them.

8.4. Improving Link Ethiopia Monitoring and Evaluation

The educational outcomes identified in this paper, given that they reflect both the relevant measures that allow impact to be compared across educational projects, and the nature of the available data in Ethiopian schools, could be considered as key performance indicators to be applied across Link Ethiopia projects. If Link Ethiopia does collect data on random students across the schools of sponsored students, this data can be used to build up a picture of how the different schools perform across these indicators. Alternatively, for each class the rosters have a summary sheet, and this information could also be used, rather than basing estimates on randomly selected students.

The benefit of collecting this data is twofold. Firstly, it may help Link Ethiopia to build up a picture of the different schools, and help identify schools with a greater need that perform less well, or alternatively identify well performing schools where new programmes will have a greater impact. Secondly, when Link Ethiopia do carry out a programme or project at a school level such an approach will allow them to measure the impact of the programme. Measuring the change in these indicators in the project school, as opposed to a control school where the project did not take place, using a difference-in-difference approach, can help the organisation to understand the impact of their projects on the schools.

As this paper has repeatedly stated prospective analysis is largely more accurate than retrospective analysis in measuring impact. When looking at historic projects, there may be too many unknown variables impacting outcomes at a school-level to carry out meaningful analysis on these. When introducing new projects though, whether it's a water station or teacher-training, identifying the key outcome variables in advance, and suitable control schools, and control variables to account for other school-level changes over the project period, and using difference-in-difference

analysis can be part of a robust monitoring and evaluation policy, applied across Link Ethiopia projects.

Whether evaluations are prospective or retrospective, the real strength of using the educational outcomes used in this study, is that such an approach will involve using data that is generally available across all schools, and a consistent approach to evaluation will allow Link Ethiopia to compare the impact of different projects, and thus learn which projects are more effective than others. As has been a consistent theme throughout this approach, impact analysis gains added meaning when considered in terms of cost-effectiveness, and thus these outcomes should always be considered in relation to the expenditure on the projects. Given the use of project budgets, cost-effectiveness metrics should be straightforward to calculate.

8.5. Conclusions for Link Ethiopia

The average increase per student of 0.31 years of schooling, should be considered a success by Link Ethiopia. At £391 per additional year of schooling the scheme is less cost-effective than some comparable schemes. While the data suggests that up to the average sponsorship duration of 46.2 months, promotion and enrolment rates of sponsored students have not improved overall, the data predicts that Link can expect to see significant improvements in these indicators from now on. The non-linear relationship between duration of sponsorship and promotion and enrolment warrants serious consideration by the organisation, particularly to identify the cause of the declining trends in the first years of sponsorship. There was no quantitative evidence for CCTs in addition to school resources improving educational outcomes, however this may have been due to data limitations, and therefore this should be a top priority for future research, especially as the qualitative data supported the effectiveness of CCTs.

The organisation needs focus on how it can improve its student selection, and how it can support students in the first years of sponsorship, especially by expanding the emotional support provided to students. The most effective aspects of the programme are the provision of exercise books and uniforms, and Link Ethiopia needs to improve the functioning of its sponsorship clubs, but has real scope to

achieve this and other gains from merging the Kindu and Link Ethiopia sponsorship programmes. Finally, Link Ethiopia can use this report to build a robust monitoring and evaluation system for both the sponsorship programme and across the organisation's other programmes.

Bibliography

- Behrman, J.R., Parker, S.W. & Todd, P.E., 2011. Do school subsidy programs generate lasting benefits? A five-year follow-up of Oportunidades participants. *Journal of Human Resources*, 46(1), pp.93–122.
- Behrman, J.R., Sengupta, P. & Todd, P., 2005. Progressing through PROGRESA: An Impact Assessment of a School Subsidy Experiment in Rural Mexico. *Economic Development and Cultural Change*, 54(1), pp.237–275.
- Deaton, A., 2010. Instruments, Randomization, and Learning about Development. *Journal of Economic Literature*, 48(2), pp.424–455.
- van Eekelen, W., 2013. Revisiting child sponsorship programmes. *Development in Practice*, 23(4), pp.468–480. Available at: http://dx.doi.org/10.1080/09614524.2013.790936.
- Evans, D. & Ghosh, A., 2008. *Prioritizing educational investments in children in the developing world, Working Paper*, Santa Monica.
- Evans, D., Kremer, M. & Ngatia, M., 2009. *The Impact of Distributing School Uniforms on Children's Education in Kenya*, Available at: http://www.poverty-action.org/project/0213.
- Filmer, D. & Schady, N., 2008. Getting Girls into School: Evidence from a Scholarship Program in Cambodia. *Economic Development and Cultural Change*, 56(3), pp.581–617.
- GiveWell, 2009. Developing-world education (in-depth review). Available at: http://www.givewell.org/international/education [Accessed January 11, 2016].
- Glewwe, P. et al., 2004. Retrospective vs. prospective analyses of school inputs: The case of flip charts in Kenya. *Journal of Development Economics*, 74, pp.251–268.
- Glewwe, P., Kremer, M. & Moulin, S., 2009. Many Children Left Behind? Textbooks and Test Scores in Kenya. *American Economic Journal: Applied Economics*, 1(1), pp.112–135.
- Glewwe, P., Ross, P.H. & Wydick, B., 2013. *Developing Aspirations : The Impact of Child Sponsorship on Self-Esteem and Life Expectations, Unpublished Working Paper*, Available at: http://repository.usfca.edu/econ/9.
- Gretland, E., Robinson, B. & Baig, T., 2014. *An Investigation into the Barriers to Female Education in Ethiopia*, Gondar.
- Handa, S. & Peterman, A., 2007. Child Health and School Enrollment A Replication. *The Journal of Human Resources*, 42(4), pp.863–880. Available at: http://www.jstor.org/stable/40057332 [Accessed July 18, 2016].

- Jennings, M. & Poppe, R., 2012. Education Management Information System Dropout Study 1999-2003 E.C. (2006/07 - 2010/11 G.C.)., (July).
- Kingdon, G. & Drèze, J., 2001. School Participation in Rural India. *Review of Development Economics*, 5(1), pp.1–24. Available at: http://dx.doi.org/10.1111/1467-9361.00103.
- Kremer, M., Miguel, E. & Thorton, R., 2009. Incentives to Learn. *The Review of Economics and Statistics*, 91, pp.437–456.
- Kremer, M., Moulin, S. & Namunya, R., 2003. Decentralization: A Cautionary Tale. *Poverty Action Lab*, (10).
- Link Ethiopia, 2016. About Us. Available at: www.linkethiopia.org/about-us [Accessed July 31, 2016].
- Link Ethiopia, 2015a. Annual Report 2014/15, London.
- Link Ethiopia, 2015b. Sponsorship Guidelines, London.
- MacAskill, W., 2015. *Doing Good Better*, London: Guardian Faber Publishing.
- Miguel, E. & Kremer, M., 2004. Worms: Identifying Impacts on Education and Health in the Presence of Treatment Externalities. *Econometrica*, 72(1), pp.159–217.
- Ministry of Education, 2015a. *Education Sector Development Programme V (ESDP V)*, Addis Ababa.
- Ministry of Education, 2015b. *Education Statistics Annual Abstract 2009 E.C.* (2013/14), Addis Ababa.
- Pettit, J. & Shutt, C., 2008. *The Development Impact of Child Sponsorship: the effects of Plan International's sponsorship processes on children, their families, communities and sponsors*, Brighton.
- Rabbitts, F., 2012. Child sponsorship, ordinary ethics and the geographies of charity. *Geoforum*, 43(5), pp.926–936. Available at: http://dx.doi.org/10.1016/j.geoforum.2012.04.006.
- Robinson, B., 2016. *Designing an Impact Evaluation of Link Ethiopia's Child Sponsorship Programme. Unpublished Manuscript, University of Copenhagen,*
- Schultz, T.P., 2004. School subsidies for the poor: evaluating the Mexican Progresa poverty program. *Journal of Development Economics*, 74(1), pp.199–250. Available at: http://linkinghub.elsevier.com/retrieve/pii/S0304387803001858.
- Singer, P., 2015. *The Most Good You Can Do: How Effective Altruism Is Changing Ideas About Living Ethically*, New Haven: Yale University Press.
- Stalker, P., 1982. Please do not sponsor this child. *New Internationalist*, 111, pp.7–28.

The Kindu Trust, 2015. Annual Report 2014/15, London.

- The Kindu Trust, 2016a. Child Sponsorship. Available at: https://kindutrust.org/childsponsor/ [Accessed June 20, 2016].
- The Kindu Trust, 2016b. Our History. Available at: https://kindutrust.org/history/ [Accessed July 20, 2016].

The Kindu Trust, 2014. Rules and Policies for the Sponsorship Programme, London.

UNESCO, 2015. Education for All 2015 National Review Report: Ethiopia.

- Vermeersch, C. & Kremer, M., 2004. School Meals, Educational Achievement, and School Competition: Evidence from a Randomized Evaluation. *SSRN Electronic Journal*. Available at: http://www.ssrn.com/abstract=667881.
- Watson, B., 2014a. A Typology of Child Sponsorship Activity. In B. Watson & M. Clarke, eds. *Child Sponsorship: Exploring Pathways to a Brighter Future*. Basingstoke: Palgrave Macmillan, pp. 41–65.
- Watson, B., 2014b. Origins of Child Sponsorship: Save the Children Fund in the 1920s. In B. Watson & M. Clarke, eds. *Child Sponsorship: Exploring Pathways to a Brighter Future*. Basingstoke: Palgrave Macmillan, pp. 18–40.
- Watson, B. & Clarke, M., 2014. Introduction to Key Issues in Child Sponsorship. In B. Watson & M. Clarke, eds. *Child Sponsorship: Exploring Pathways to a Brighter Future*. Basingstoke: Palgrave Macmillan, pp. 1–17.
- Watson, B., Lockton, H. & Pawar, M., 2014. Issues in Historic Child Sponsorship. In
 B. Watson & M. Clarke, eds. *Child Sponsorship: Exploring Pathways to a Brighter Future*. Basingstoke: Palgrave Macmillan, pp. 66–95.
- Weldon, A., 2015. Ethiopia Launches National Deworming Programme Targeting Children. Available at: http://www3.imperial.ac.uk/newsandeventspggrp/imperialcollege/centres/sci/new ssummary/news_7-1-2015-16-56-37 [Accessed January 11, 2016].
- Wydick, B., Glewwe, P. & Rutledge, L., 2013. Does International Child Sponsorship Work? A Six-Country Study of Impacts on Adult Life Outcomes. *Journal of Political Economy*, 121(2), pp.393–436. Available at: http://www.jstor.org/stable/info/10.1086/670138.

Appendix A: Interview Guide – Sponsored Students

About school

- What do you think about your school?
- How do you get on with your teachers?
- What is your relationship like with other students?
- What do you do with your time when you are not in school?
- What subjects do you like?
- Why do you like those subjects
- What do you find hard about school?
- Who decides in your family if you attend school?
 - Is it you, or your parents?

Life Story and Present Situation

- Tell me about what it was like when you were very young?
 - How many brothers and sisters do you have?
 - Are they older or younger than you?
 - Which house did you grow up in?
 - What jobs did your parents do? Do they still do those jobs?
- When did you start going to school?
 - How old were you?
 - o Did your brothers and sisters attend school?
 - Tell me about (year/grade before sponsorship) do you remember what happened that year?
 - And tell me about the year after...
- How did you become to be sponsored?
 - Were you aware you were on a waiting list for sponsorship?
 - \circ $\,$ Who told you were being sponsored and what that meant?
 - Did you know about Link Ethiopia and sponsorship before you were sponsored?
 - \circ Did you know of students at the school who were sponsored?
 - Did you talk about being sponsored with your family?

About the sponsorship scheme

- Which bits of the support do you find most helpful to your education?
- What changes to the programme would you recommend?
- Do you talk about sponsorship with your family?
- Do you talk about being sponsored with your friends?
- Do you know the other children in the school who are sponsored? Are they friends of yours?
- Why are you happy to be part of the Link Ethiopia programme?
- What do you think would be different if you hadn't been sponsored by Link Ethiopia? Do you think you would still be in school?

Appendix B: Interview Guide – Link Coordinator

- Tell me about the Link sponsorship scheme
 - How does it work?
 - What support do students receive?
 - o Tell me about the resources Link provides the children with?
 - How are the children selected? What do you think of the selection process?
- Tell me about the Sponsorship Club?
- What do you think makes the difference for children to stay in school? Do you think the Link resources help them to stay in school?
- In school, do the sponsored children become friends with each other?
- Do you think the children are happy to be part of the Link Ethiopia programme?
- Do the other children know which students are sponsored? What do they think of this?
- What do you think would happen to the children if they didn't receive sponsorship?
- What more could Link do to support the children?
- Do you know of other sponsorship schemes in your school? How do they compare?
- Is there anything you can do to support the student?
- Do you know of any students on the scheme who have dropped out? What do you think the reasons for dropping out are?

Appendix C: Interview Ethics

Orally shared at the beginning of each interview

I am student from the UK, doing my master's thesis studying the impact of sponsorship in Ethiopia. I just wanted check that:

- You are happy to take part in this interview
- That anything you say will be reported anonymously
- Whatever you say in the interview will not be shared with Link Ethiopia or The Kindu Trust, and will in no way effect the support you receive from Kindu/Link Ethiopia
- That you are happy for the interview to be audio recorded.

Safeguarding – Link Ethiopia or Kindu staff member to be present

Enrolment	Model 1	Model 2	Model 3	Model 4
т	0	0	0	0
	(.)	(.)	(.)	(.)
Gender	0	0	0	0
	(.)	(.)	(.)	(.)
Age=5	0	0	0	0
	(.)	(.)	(.)	(.)
Age=6	-0.000176	-0.00659	-0.0144	-0.00694
	(0.997)	(0.890)	(0.756)	(0.885)
Age=7	-0.0637	-0.0702	-0.0853	-0.0653
	(0.237)	(0.186)	(0.085)	(0.209)
Age=8	-0.163*	-0.169*	-0.184**	-0.162*
	(0.024)	(0.020)	(0.009)	(0.026)
Age=9	-0.306***	-0.313***	-0.328***	-0.305***
	(0.000)	(0.000)	(0.000)	(0.000)
Δge=10	-0.426***	-0.434***	-0.450***	-0.427***
	(0.000)	(0.000)	(0.000)	(0.000)
Age=11	-0.510***	-0.520***	-0.537***	-0.512***
	(0.000)	(0.000)	(0.000)	(0.000)
Age=12	-0.620***	-0.630***	-0.648***	-0.621***
Ag0=12	(0.000)	(0.000)	(0.000)	(0.000)
Age=13	-0.726***	-0.736***	-0.753***	-0.726***
	(0.000)	(0.000)	(0.000)	(0.000)
Age=14	-0.881***	-0.891***	-0.909***	-0.883***
	(0.000)	(0.000)	(0.000)	(0.000)
Age=15	-0.986***	-0.997***	-1.014***	-0.991***
	(0.000)	(0.000)	(0.000)	(0.000)
Age=16	-1.129^^^	-1.141^^^	-1.15/^^^	-1.139***
	(0.000)	(0.000)	(0.000)	(0.000)
Age=17	-1.351^^^	-1.362***	-1.379^^^	-1.363***
	(0.000)	(0.000)	(0.000)	(0.000)
Age=18	-1.764***	-1.//6***	-1./94^^^	-1.770***
	(0.000)	(0.000)	(0.000)	(0.000)
Age=19	-2.066***	-2.078***	-2.098***	-2.069***
	(0.000)	(0.000)	(0.000)	(0.000)
Age=20	-2.261	-2.2/3	-2.294	-2.260
	(0.000)	(0.000)	(0.000)	(0.000)
Age=21	-2.939	-2.951	-2.973	-2.935
	(0.000)	(0.000)	(0.000)	(0.000)
Age=22	-3.120 ^{mm}	-0.100	-3.100	-3.118
	(0.000)	(0.000)	(0.000)	(0.000)
Year=2006	U	U ()	U ()	U ()
	(.)	(.)	(.)	(.)
Year=2007	0.142	0.143	0.145	0.141
	(0.078)	(0.077)	(0.071)	(0.088)

Appendix D: Example Regression with Student Fixed Effects

Year=2008	0.144*	0.142*	0.138*	0.144*
	(0.023)	(0.026)	(0.028)	(0.026)
No 0000	0.0722	0.0700	0.0655	0.0718
Year=2009	(0.199)	(0.216)	(0.240)	(0.211)
Year=2010	0.0728	0.0724	0.0679	0.0755
	(0.128)	(0.133)	(0.150)	(0.122)
Year=2011	0.0725	0.0705	0.0653	0.0734
	(0.068)	(0.078)	(0.093)	(0.069)
Year=2012	0.0614	0.0618	0.0577	0.0646
	(0.062)	(0.064)	(0.077)	(0.055)
Year=2013	0.0408	0.0418	0.0392	0.0453
	(0.144)	(0.137)	(0.159)	(0.112)
Year=2014	-0.00554	-0.00442	-0.00627	-0.00120
	(0.799)	(0.841)	(0.773)	(0.957)
Year=2015	0	0	0	0
	(.)	(.)	(.)	(.)
Orreade d	0	0	0	0
Grade=1	(.)	(.)	(.)	(.)
Orreade 0	0.0828**	0.0820**	0.0816**	0.0809**
Grade=2	(0.005)	(0.006)	(0.007)	(0.007)
0	0.239***	0.238***	0.235***	0.238***
Grade=3	(0.000)	(0.000)	(0.000)	(0.000)
O to to t	0.368***	0.366***	0.361***	0.369***
Grade=4	(0.000)	(0.000)	(0.000)	(0.000)
Orrende E	0.484***	0.483***	0.476***	0.486***
Grade=5	(0.000)	(0.000)	(0.000)	(0.000)
Grade=6	0.639***	0.639***	0.631***	0.646***
	(0.000)	(0.000)	(0.000)	(0.000)
Grade=7	0.741***	0.743***	0.736***	0.746***
	(0.000)	(0.000)	(0.000)	(0.000)
Grade=8	0.906***	0.906***	0.898***	0.910***
	(0.000)	(0.000)	(0.000)	(0.000)
Grade=9	0.107	0.102	0.0662	0.104
	(0.396)	(0.421)	(0.593)	(0.409)
Grade=10	0.313*	0.309*	0.273*	0.303*
	(0.023)	(0.027)	(0.045)	(0.028)
Orreade 44	0.589***	0.586***	0.550**	0.575***
Grade=11	(0.001)	(0.001)	(0.001)	(0.001)
Grade=12	0.999***	0.996***	0.959***	0.989***
	(0.000)	(0.000)	(0.000)	(0.000)
Oshaal 1	0	0	0	0
School=1	(.)	(.)	(.)	(.)
O the set of	-0.872***	-0.877***	-0.895***	-0.843***
Scnooi=2	(0.000)	(0.000)	(0.000)	(0.000)
	0.947***	0.952***	0.988***	0.905***
School=3	(0.000)	(0.000)	(0.000)	(0.000)
School=4	0.197	0.194	0.198	0.190
	(0.407)	(0.412)	(0.409)	(0.397)

School=5	0	0	0	0
	(.)	(.)	(.)	(.)
School=6	-0.822***	-0.829***	-0.868***	-0.783***
	(0.000)	(0.000)	(0.000)	(0.000)
0.4	0	0	0	0
School=7	(.)	(.)	(.)	(.)
School=8	-0.0302	-0.0295	-0.0383	0.00307
	(0.258)	(0.275)	(0.177)	(0.928)
School=9	0.198	0.191	0.197	0.221
	(.)	(.)	(1.000)	(1.000)
School=10	0.957***	0.963***	1.003***	0.912***
	(0.000)	(0.000)	(0.000)	(0.000)
School-11	0.748***	0.756***	0.798***	0.687***
SC1001=11	(0.000)	(0.000)	(0.000)	(0.000)
Sabaal-12	0.899***	0.908***	0.937***	0.936***
0011001=12	(0.000)	(0.000)	(0.000)	(0.000)
School-13	0	0	0	0
	(.)	(.)	(.)	(.)
School=14	0	0	0	0
	(.)	(.)	(.)	(.)
School=15	0	0	0	0
	(.)	(.)	(.)	(.)
School=16	-1.100***	-1.103***	-1.104***	-1.129***
	(0.000)	(0.000)	(0.000)	(0.000)
School=17	0	0	0	0
	(.)	(.)	(.)	(.)
School=18	0	0	0	0
	(.)	(.)	(.)	(.)
School=19	0	0	0	0
	(.)	(.)	(.)	(.)
S₁	-0.0395			
-	(0.085)			
S ₂		-0.00302		
		(0.199)	0.000.170	0.00050*
S₃			-0.000472	-0.00356^
-			(0.509)	(0.017)
$(S_3)^2$				0.0000499^
-	4.040	4 007	4.050	(0.029)
Constant	1.218	1.227	1.250	1.216
	(.)	(.)	(1.000)	(1.000)
Observations	1219	1219	1219	1219

p-values in parentheses p < 0.05, p < 0.01, p < 0.001